

Domestic and External Sovereign Debt

Paola Di Casola and Spyridon Sichlimiris

November 2017

WORKING PAPERS ARE OBTAINABLE FROM

www.riksbank.se/en/research

Sveriges Riksbank • SE-103 37 Stockholm Fax international: +46 8 21 05 31 Telephone international: +46 8 787 00 00

The Working Paper series presents reports on matters in the sphere of activities of the Riksbank that are considered to be of interest to a wider public. The papers are to be regarded as reports on ongoing studies and the authors will be pleased to receive comments.

The opinions expressed in this article are the sole responsibility of the author(s) and should not be interpreted as reflecting the views of Sveriges Riksbank.

Domestic and External Sovereign Debt*

Paola Di Casola Spyridon Sichlimiris[†] Sveriges Riksbank Örebro University

Sveriges Riksbank Working Paper Series No. 345 November 2017

Abstract

Why do countries tend to repay their domestic and external debt, even though the legal enforcement of the sovereign debt contract is limited? Contrary to conventional wisdom, we argue that temporary market exclusion after default is costly. When the domestic financial market is characterized by a scarcity of private saving instruments, a government can partition its debt market into domestic and external segments, by restricting capital flows, to exploit its market power. The government's market power mitigates the problem of limited commitment, by making default a more costly option. Consequently, it extends the government's external debt capacity. We replicate the domestic and external sovereign debt for non-advanced economies, by unveiling their link to financial repression.

Keywords: sovereign debt, sovereign default, financial repression, financial development, capital controls.

JEL codes: E21, E44, E60, F34, F38, G15, G18, H63, O16.

[†]Corresponding author: Spyridon Sichlimiris, spyridon.sichlimiris@oru.se.

^{*}We are indebted to Tore Ellingsen and Lars Ljungqvist for their guidance since the early versions of this paper. We are also grateful to Klaus Adam, Christoph Bertsch, Tobias Broer, Ana Maria Ceh, Giovanni Dell'Ariccia, David Domeij, Martial Dupaigne, Aitor Erce, Luca Facchinello, Daria Finocchiaro, Patrick Féve, Francois Geerolf, Nicola Gennaioli, Jungsuk Han, Christian Hellwig, Tom Krebs, Keith Kuster, Taneli Mäkinen, Björn Ohl, Franck Portier, Kjetil Storesletten, Jean Tirole, Karl Walentin, Johanna Wallenius and seminar participants at Stockholm School of Economics, Stockholm University, Toulouse School of Economics, Mannheim University, Swedish House of Finance, HEC Lausanne, SUDSWEC conference 2013, Computing in Economics and Finance 2014, 2nd Macro Banking and Finance Workshop, Nationell konferens i nationalekonomi 2014, XXXIX Simposio de la Asociación Española de Economía, RES PhD Meetings 2015, 19th Conference "Theories and Methods in Macroeconomics", Greater Stockholm Macro Group, Örebro University and Sverige Riksbank for useful discussions and comments. Financial support from the Jan Wallander and Tom Hedelius Foundation is gratefully acknowledged. The opinions expressed in this article are the sole responsibility of the author and should not be interpreted as reflecting the views of Sveriges Riksbank. All remaining errors are our own.

Sovereign debt is characterized by limited legal enforcement. As a consequence, countries have defaulted on their domestic and external debt throughout their history, as shown in Reinhart and Rogoff (2011).¹ However, sovereign defaults are rare and the domestic and external sovereign debt are substantial. In Figure 1 we utilize an updated version of Panizza (2008)'s dataset of government debt. The dataset distinguishes domestic and external debt according to the place of issuance and the legal jurisdiction. Reinhart and Rogoff (2011) mention that in most countries, for most of their history, the jurisdiction has coincided with the currency of the debt and the identity of the bondholders.² We classify the non-advanced economies according to the World bank Income Group they belong to and plot the average domestic and external debt against the average total debt for the period 1970-2010. The average total debt ranges from 52 percent to 79 percent. The average external debt decreases with the level of economic development, whereas the average domestic debt increases with the level of economic development.

What can explain these debt levels under limited enforcement of sovereign debt contracts? Previous research has argued that, even in the absence of legal enforcement, the government may repay its debt because of the consequences of default. A prominent consequence of default is the loss of reputation, that leads to financial market exclusion (Eaton and Gersovitz, 1981). Empirically, market exclusion after default is temporary. Quantitatively, financial market exclusion alone is an insufficient punishment after default to rationalize the observable sovereign debt levels. Aguiar and Gopinath (2006) illustrate, in calculations à la Lucas (1985), that the welfare benefits from smoothing the business cycle are too low to justify the government's repayment choice, when the only punishment after default is financial market exclusion. Therefore, it has not been possible to reconcile the following two regularities that characterize the sovereign debt of non-advanced economies.

- Fact 1: Substantial domestic and external debt levels and low sovereign default frequency.
- Fact 2: Short-lived market exclusion after default.

We show that these features are consistent with a model of sovereign debt in which the

¹The authors report evidence that outright defaults on domestic debt did happen, despite the common belief that domestic debt would only be inflated away. "Why would a government refuse to pay its domestic public debt in full when it can simply inflate the problem away? One answer, of course, is that inflation causes distortions, especially to the banking system and the financial sector. Sometimes, the government may view repudiation as the lesser evil," (Reinhart and Rogoff, 2011, pag. 326-327). The limited role for inflation is also confirmed by the findings in Forslund, Lima and Panizza (2011), who fail to find a significant correlation between the inflation history and the sovereign debt composition for non-advanced economies.

²The choice of using the updated version of Panizza (2008)'s dataset is due to the statistical distinction used for domestic and external sovereign debt and the extensive country-time dimension. As mentioned in Panizza (2008), this is the most accurate way to distinguish domestic from external debt, because it is possible to find correct information on the legal jurisdiction. Similar accuracy cannot be achieved for the identity of the bondholders and the currency of the debt.



Figure 1: Average domestic and external sovereign debt as percentage of GDP for nonadvanced economies during 1970-2010. The data include 38 Low income countries, 46 Lower-Middle income countries and 35 Upper-Middle income countries (list of countries in appendix A).

Source: Own calculations based on Ugo Panizza's dataset described in Panizza (2008). *Description:* With hollow circle we represent the average domestic debt as percentage of GDP for each income group. With hollow diamond we represent the average external debt as percentage of GDP for each income group

government lacks commitment to repay its debt and the economy is characterized by financial underdevelopment, that takes the form of a scarcity of private savings instruments. A central idea is that the government exploits the market power in its own economy in order to mitigate the lack of commitment. Consequently, it also alleviates the scarcity of savings instruments. Due to the government's market power, financial autarky during default is costly. The government exercises its market power by imposing capital controls,³ so that part of the population is subject to financial repression.⁴ We find support for our mechanism in Escolano, Shabunina and Woo (2016), who show that, on average, non-advanced economies paid a negative real interest rate on their debt (minus 6 percent for the period

³The normalized index of financial liberalization constructed by Abiad, Detragiache and Tressel (2008) for the period 1973-2005 is equal to 0.42 for non-advanced economies, in contrast to 0.69 for advanced economies. Fernández, Rebucci and Uribe (2015) report the average index of capital controls for various economies for the period 1995-2011. The average value for developed economies is 0.07, versus 0.35 for emerging economies and 0.54 for low-income countries. Moreover, they find that capital controls are acyclical.

⁴We use the following definition for financial repression provided by Kirkegaard, Reinhart and Sbrancia (2011). Financial repression includes directed lending to the government by captive domestic audiences (such as pension funds or domestic banks), explicit or implicit caps on interest rates, regulation of cross-border capital movements, and (generally) a tighter connection between government and banks, either explicitly through public ownership of some of the banks or through heavy "moral suasion". Financial repression is also sometimes associated with relatively high reserve requirements (or liquidity requirements), securities, transaction taxes, prohibition of gold purchases (as in the US from 1933 to 1974), or the placement of significant amounts of government debt that is non-marketable.

1966-2010), as a result of financial repression. As highlighted by Obstfeld (1993), Dooley (1995), Aizenman (2004) and Kirkegaard, Reinhart and Sbrancia (2011), governments have tried to use financial repression through capital controls to reduce the domestic debt burden throughout history. However, what has remained unexplored is how financial repression is linked to external debt and default.

We extend Eaton and Gersovitz (1981)'s framework along two dimensions, in order to link external debt and default with domestic debt. First, we introduce a domestic financial market that is less developed compared to the international financial market. More specifically, the domestic financial market is characterized by a scarcity of savings instruments. This modelling choice allows us to introduce domestic debt in a straightforward way. Second, we introduce an additional type of agents, that are savers. As in Eaton and Gersovitz (1981), the government cares about the welfare of the agents that would like to smooth consumption fluctuations due to endowment uncertainty. Hence, the welfare of the savers is not in the government's objective function. The savers are subject to financial repression. At the heart of the model is the interaction between consumption smoothing under limited commitment and market power.

Our contribution is twofold. First, we resolve the quantitative disconnect between shortlived market exclusion and sovereign debt levels. Seminal quantitative models of external debt, such as Aguiar and Gopinath (2006), Arellano (2008) and the subsequent literature, absent any other cost of default (such as endogenous or exogenous output costs), would predict insignificant debt levels. Thus, creating a stunning puzzle: the disconnect between market exclusion and sovereign debt levels. A reason behind this failure is that the welfare gains from smoothing shocks are small, as already pointed out by Aguiar and Gopinath (2006). Contrary to that, we claim that financial market exclusion is a costly punishment after default. The reason is that the government can benefit from segmenting its debt market. Importantly, the domestic debt market is a source of cheap funding for the government, because it can be subject to financial repression. We illustrate in a quantitative manner that temporary exclusion from segmented debt markets is a sufficient punishment after default to rationalize the observable sovereign debt levels and the composition of debt. We calibrate the model to three countries (Argentina, Mexico and Russia) that defaulted jointly on their domestic and external debt and used financial repression policies in the past decades. Consistently across the three countries, the model predicts short-lived market exclusion in order to account for the evidence on domestic and external sovereign debt.

The optimal behavior from the model is mostly determined by the interaction of the government's motive to smooth consumption in the presence of limited commitment and its market power in the domestic economy. The market power makes default a worse

alternative compared to debt. Therefore, even a seemingly mild punishment after default, such as temporary financial market exclusion, is enough for our model to replicate realistic debt levels. The issue that remains open is whether the market power should be stronger in booms or recessions. If we want to describe the behavior of non-advanced economies, we need the market power to be relatively stronger in booms.⁵ Thus, across the cycle, the scarcity of savings instruments is more pronounced in booms. Then, default incentives are stronger in recessions, which make default more likely in those times, in line with the evidence. Domestic debt is relatively more valuable than external debt in booms, due to the procyclical market power. Given the tension between smoothing fluctuations through debt issuance and procyclical market power, external debt may be higher or lower in booms, depending on the government's patience, the volatility and the persistence of the output process. For the three countries studied, the optimal behavior prescribes that, conditional on debt to be repaid, in bad times the domestic debt share is lower and defaults are more likely. Hence, without an a priori preference for domestic bondholders, the government tends to have lower domestic debt when default incentives are higher.

The second contribution of our paper is a unified explanation for the evidence on sovereign debt for non-advanced economies, as presented in Figure 1; it is a tale of financial repression. We obtain a negative real interest rate on government debt of similar magnitude as the one calculated by Escolano, Shabunina and Woo (2016) for non-advanced economies. Through the lens of our model, the Low Income group is able to sustain more sovereign debt than the Lower-middle Income and the Upper-middle Income groups, because the scarcity of private savings instruments is more pronounced. That implies that the government's market power for the Low Income group is higher than for the other two groups. Domestic debt is more valuable when the market power is higher. However, high market power implies that the marginal benefit from the additional debt issuance decreases faster. These two effects imply that the Low Income group can sustain higher total debt levels, mostly as external debt. The other two groups feature a more balanced composition of debt and lower total debt levels, because limited commitment is more prevalent.

Related literature. Our model builds on the seminal model of external sovereign debt in Eaton and Gersovitz (1981), and its quantitative implementation in Arellano (2008), Aguiar and Gopinath (2006). There is a vast literature that departed from there and developed in different directions (see Aguiar et al. (2016), D'Erasmo, Mendoza and Zhang (2016) for recent literature reviews). Our contribution to the sovereign debt literature is that we make financial market exclusion quantitatively relevant punishment after default. We are closer

⁵For the curious reader, this is mainly achieved by assuming that the same endowment shock hits the different types of agents in the small open economy. In addition, we assume that capital controls are acyclical, in line with the evidence in Fernández, Rebucci and Uribe (2015).

to the papers that introduced domestic debt, such as Mallucci (2015), Perez (2015) and D'Erasmo and Mendoza (2016), who also assume non-discriminatory default for domestic and external debt. Mallucci (2015) and Perez (2015) include a domestic financial sector and focus on the negative effect of the sovereign default on domestic banks. Their mechanism is based on the theoretical contribution in Gennaioli, Martin and Rossi (2014), who show how the development of domestic financial institutions affect the governemnt's default incentives, due to the effect of default on the level of credit in the domestic economy. D'Erasmo and Mendoza (2016) build on the idea that default can be used by the government to redistribute resources among the domestic agents, by assuming that the households enter the government's utility with a weight given by their wealth. Hence, all these papers focus on some form of internal cost of default and assume that the government has a preference for domestic bondholders. They follow from the interaction between the government's market power and its consumption smoothing motive. Our model is consistent with the evidence on financial repression in non-advanced economies.

To the best of our knowledge, our model is the first one to integrate the study of domestic and external debt and default with the issue of financial repression, in the form of market segmentation, and to show the quantitative implications of this setting. Two recent non-quantitative papers have addressed the connection between financial repression and debt, Acharya and Rajan (2013) and Chari, Dovis and Kehoe (2016). However, financial repression in these papers takes the form of taxation that incentivizes banks to hold domestic debt, which means that it is not based on market segmentation. Guidotti and Kumar (1991) and Giovannini and de Melo (1993) report evidence of a positive gap between international and domestic interest rates on government debt of developing countries for the 70's and 80's. More recently, Du and Schreger (2016) find a positive gap between the spread of local currency debt and foreign currency debt of emerging economies. Given this evidence on price discrimination, connecting financial repression and market segmentation appears more relevant. In Acharya and Rajan (2013) the existence of external debt is due to the non-discriminatory default causing a loss of resources to redistribute internally for the (populist) government. Instead, in Chari, Dovis and Kehoe (2016) the assumption of complete markets rules out default in equilibrium, hence the mechanism allowing for debt in equilibrium is different from ours.

Outside the sovereign debt literature, a macro-finance paper that has implemented the idea of government's market power and segmented financial market is Gordon and Li (2003). They show that once the government has market power in the domestic market and domestic investors' demand for assets is less elastic than external investors' demand for

assets, segmenting the market is the optimal choice. We contribute to the macro-finance literature by arguing that the government's market power can alleviate the limited commitment problem that characterizes government debt.

Outline. The paper continues as follows. Section 1 describes the model environment and the recursive equilibrium. Section 2 shows the quantitative relevance of financial market exclusion. In section 3 we unveil the link between financial repression and sovereign debt for non-advanced economies. Section 4 discusses the role of the model's key assumptions and section 5 concludes.

1 Model

The world economy consists of a small open endowment economy and the rest of the world. The time of the model is discrete and the horizon is infinite. We enrich the standard external sovereign debt and default model - á la Aguiar and Gopinath (2006) and Arellano (2008) - with a domestic financial market that is less financially developed than the international financial market. For insurance purposes, the government is allowed to issue domestic debt in addition to external debt and default on it. In contrast to the above mentioned models, the only punishment after default is short-lived market exclusion.

The small open economy is populated by two types of agents: risk-averse domestic consumers and risk-neutral domestic bondholders. Both types of agents face the same endowment uncertainty. A government provides insurance to the risk-averse consumers against endowment shocks. It does so by issuing one-period non-contingent domestic and external bonds. Domestic bondholders have a pro-cyclical inelastic demand for assets (government and private bonds). In addition, the supply of domestic private saving instruments is limited. The government possesses market power in its own economy, because it can affect the supply of savings instruments. Instead, the rest of the world is populated by risk-neutral agents that have access to a frictionless competitive financial market. The government restricts the access of domestic bondholders to the international financial market in order to exploit its market power. Thus, domestic bondholders are financially repressed in favor of domestic consumers.⁶ The government issues domestic debt that is less costly compared to the external debt. The fact that the government has market power in the domestic economy compared to not having market power in the international financial market implies that both types of debt are being traded in equilibrium. Due to the assumption of incom-

⁶In our model the government is not a benevolent social planner for the whole population. This view of the government is in line with studies from political science, such as Menaldo (2016). The author argues that in countries with weak state capacity, where it is difficult to collect taxes, governments use financial repression to distort the market. In this way, they can raise revenues for their fiscal needs and to support coalitions to maintain power.

plete markets, default arises as an equilibrium outcome. The government faces the same punishment for defaulting in one market or the other (temporary financial autarky), hence only non-selective default is optimal in equilibrium. Our assumption is that by damaging the credit relationship in one debt market, the government damages its credit relationship also in the other debt market. These spillovers effects are similar to Cole and Kehoe (1998). If the government discriminates against external bondholders, it may violate international regulations or norms, thus entailing harsh punishment. External bondholders can retaliate in various ways against the defaulting economy in case of differential treatment. In our model we require that selective sovereign defaults on external debt are too costly for the government. Therefore, we focus on non-selective defaults.⁷

We proceed by describing the agents that populate the economy and the recursive equilibrium.

1.1 Agents

1.1.1 Domestic Consumers - Government

There is a mass λ of identical, infinitely lived consumers, who are risk averse. Every period, each domestic consumer receives an endowment *y* that is stochastic and not storable. We consider the case where the endowment follows a Markov process. Domestic consumers trade across the financial markets through the government. The government taxes their endowment in lump-sum fashion and provides insurance to the domestic consumers by issuing one-period non-contingent domestic and external government bonds. It can ex-ante identify the bondholders, but it cannot commit to repay the debt back to any of them. We model the lack of commitment through the strategic default choice. Formally, the problem of the government is

$$V(Y^G, B) = \max\{V^d(Y^G), V^{nd}(Y^G, B)\},\tag{1}$$

where $Y^G = \lambda y$ denotes the government's endowment and *B* the total debt to be repaid. In each period the government chooses whether to default or issue domestic and external debt by comparing the value of default, $V^d(Y^G)$, and the value of not defaulting, $V^{nd}(Y^G, B)$. The state variables of the problem are the current endowment and the total amount of debt to be repaid, because the punishment after default is independent of the identify of the bondholder. The value of defaulting $V^d(Y^G)$ is defined as

⁷in Section 4 we discuss the role of this assumption. The main mechanism of the model, that relies on market power and limited commitment, would still hold with discriminatory default.

$$V^{d}(Y^{G}) = u(Y^{G}) + \beta \mathbb{E}\left[\theta V(Y^{G}, 0) + (1 - \theta)V^{d}(Y^{G})\right].$$
(2)

If the government defaults in the current period, it receives the utility associated with the endowment shock. During the next period, with probability θ it re-enters the financial markets with zero past debt. With probability $1 - \theta$ it remains in financial autarky. The probability θ is time-invariant. In case that the government does not default, it obtains the value from trading across markets $V^{nd}(Y^G, B)$, which is defined as

$$V^{nd}(Y^G, B) = \max_{B'^D, B'^E} \{ u(Y^G - B + q^D(B', B'^D, Y^G)B'^D + q^E(B', Y^G)B'^E) + \beta \mathbb{E}V(Y'^G, B') \}$$
(3a)

subject to

$$B' = B'^D + B'^E, ag{3b}$$

where B'^D and B'^E are the quantities of the domestic and external bonds that mature in the next period and $q^D(B', B'^D, Y^G)B'^D$, $q^E(B', Y^G)B'^E$ are the revenues from the domestic and external debt issuance, respectively. Positive values of B'^D and B'^E indicate borrowing. In each period the government repays its past debt *B* and issues new domestic and external debt, affecting both the current utility and the continuation value.

1.1.2 Domestic Bondholders

We aim to generate a scarcity of private savings instruments in the domestic economy. On the supply side, we assume a fixed amount of private savings instruments. On the demand side, we create a procyclical inelastic demand for savings instruments, by adopting a two-period overlapping generations modelling structure.⁸

There is a mass $1 - \lambda$ of risk-neutral domestic bondholders who live for two periods: in the first period they are young, in the next period they are old. They receive a stochastic non-storable endowment *y* when young, but they cannot consume it. When young, they use their endowment to buy domestic government bonds b'^{D} and domestic private bonds l'. When old, they consume the expected return from their portfolio. Formally, the problem

⁸A similar modelling choice to generate a scarcity of savings instruments is presented in Farhi and Tirole (2012). The use of overlapping generations does not imply that a period in the model represents 25 years. A similar use of overlapping generations inside a macroeconomic model with infinite horizon households is offered by Clerc et al. (2015). It is beyond the scope of the paper to provide a microfounded framework in which the inelastic demand for assets emerges.

of the domestic bondholder is

$$V^{D}(Y^{G}, B', B'^{D}) = \max_{b'^{D}, l'} \{\delta(B', Y^{G})l' + (1 - \delta(B', Y^{G}))(l' + b'^{D})\}$$
(4a)

subject to

$$y = q^{D}(B', B'^{D}, Y^{G})b'^{D} + q^{l}(B', B'^{D}, Y^{G})l'.$$
(4b)

The domestic bondholder maximizes the next period's expected consumption $\delta(B', Y^G)l' + (1 - \delta(B', Y^G))(l' + b'^D)$, by taking the probability of sovereign default, $\delta(B', Y^G)$, as given. Each domestic bondholder is infinitesimally small and does not internalize how her demand for bonds influences the price of domestic government bonds $q^D(B', B'^D, Y^G)$ and the price of domestic private bonds $q^l(B', B'^D, Y^G)$. While the domestic consumers' endowment can be directly taxed by the government and used as collateral to issue debt, the domestic bondholders' endowment cannot be directly taxed. Financial repression is a form of indirect taxation.

1.1.3 External Bondholders

We model the external bondholders as identical, infinitely lived, risk-neutral agents as in Eaton and Gersovitz (1981)'s framework and its subsequent quantitative implementations by Aguiar and Gopinath (2006) and Arellano (2008). They have access to a competitive international financial market where they can trade as much as needed of external risk-free assets. These assets pay one unit of good in the following period. The external investors can also trade external government bonds. They can observe the endowment of the small open economy when it is realized. The representative external investor buys external government bonds, b'^E , as long as their expected return, $1/q^E(B', Y^G)$, is equal to the return of the risk-free asset, r^z .

1.1.4 Financial Markets Development

The domestic financial market is assumed to be less developed than the international financial market in the sense that it cannot supply enough savings instruments. The supply of savings instruments (private bonds) is limited to an exogenous amount *L*. In both markets all participants take prices as given, except that in the domestic financial market the government has market power. Eventually, the government influences the effective supply of savings instruments in the domestic financial market. These features give rise to an endogenous interest rate in the domestic financial market. The interest rate in the external market is exogenous to the small open economy.

1.1.5 Timing



The endowment shock is realized. Every agent observes the realized endowment. Afterwards, the government decides whether to default or not on the amount borrowed in the past. If the government defaults on its debt, it is in autarky in the current period. On the other hand, if the government does not default, it has access to the segmented financial markets and chooses the amount of bonds to issue in the domestic and the external market. When financial markets open, domestic investors can trade domestic government bonds and domestic private bonds. External investors can trade external government bonds and external risk-free assets. At the end of the period agents consume.

1.2 Recursive Equilibrium

Following Chatterjee et al. (2007), we define two sets which characterize the government's optimal decision to repay or to default. The repayment set is the set of values of current endowment such that, given a value of total debt *B*, the government prefers repayment to default:

$$Rep(B) = \{Y^G \in \Psi : V^{nd}(Y^G, B) \ge V^d(Y^G)\}.$$
(5)

Its complement, the default set, is the set of values of current endowment such that, given a value of total debt *B*, the government prefers default to repayment:

$$D(B) = \{ Y^G \in \Psi : V^d(Y^G) > V^{nd}(Y^G, B) \}.$$
 (6)

These sets depend only on the total amount of debt to repay at the beginning of the period, because we have simplified the problem and reduced the number of state variables in an insightful way. A government would never default selectively, because the cost for

defaulting is the same for any type of bondholder and amount of debt.⁹ Finally, for the same reason we can define the unique endogenous default probability as the probability that, given the current endowment and the total amount of bonds issued B', the next period's endowment belongs to the default set

$$\delta(B', Y^G) = \int_{D(B')} f(Y'^G, Y^G) dY'^G.$$
(7)

We can now define the maximum level of borrowing at which the default probability is zero and the maximum level of borrowing where the default event is an almost-sure event. First, the no-default-risk borrowing limit is

$$B^{MIN} = \sup\{B : D(B) = \emptyset\}.$$
(8)

Below this level government bonds behave like risk-free assets. Second, the highest level of debt that the government can attain before the probability of default is one is defined as

$$B^{MAX} = \inf\{B : D(B) = \Psi\}.$$
(9)

Within these limits the set of values of total borrowing carry a default premium paid to domestic and external bondholders.¹⁰

1.2.1 Definition of Equilibrium

A recursive equilibrium is an initial condition B_0 , pricing functions q^l , q^D , q^E , q^z , sets of value functions V, V^d , V^{nd} for the government, V^D for domestic bondholders; policy functions B'^D , B' for the government, $b^{D'}$, $b^{E'}$, l' for the bondholders; the repayment set Rep(B) and default set D(B) such that:

(*a*) Given the price functions q^D and q^E , B'^D , B', the repayment set Rep(B) and the default set D(B) satisfy the government's maximization problem (1) - (3).

(*b*) Given the price functions q^D and q^l and the government's policy functions, $b^{D'}$ and l' solve the maximization problem of the representative domestic bondholder (4a) - (4b).

(c) Given the price functions q^E and q^z and the government's policy functions, external

⁹Based on the evidence from Jeanneret and Souissi (2016), defaults on local currency debt are as likely as defaults on foreign currency debt. Thus, selective defaults appears to be of a second order importance.

¹⁰We also set a lower bound for the government's savings. This limit arises from the constraint on the consumption of domestic bondholders. We define this limit as $\underline{B}^D = -L$.

bondholders optimally choose $b^{E'}$.

(*d*) Bond markets clear, that is $\int_0^{1-\lambda} b^{D'} di = B'^D$ in case of domestic borrowing and $\int_0^1 b^{E'} di = B'^E$ in case of external borrowing. Domestic private bond markets clear, that is $\int_0^{1-\lambda} l' di = L$.

1.3 Pricing Functions

We derive the pricing functions from the optimization problems that we have defined in section 1.1. The price of the international risk-free bonds traded in the international financial markets is equal to the discount factor of the external bondholders, because the external financial market is competitive and the bondholders are risk neutral.

$$q^z = \beta^E. (10)$$

The price of the risky asset - external government bond - is such that the expected profits from its purchase are equal to the profits from the purchase of the external risk-free asset. Thus, the price of the external government bonds is a function of the external risk-free interest rate and the default probability:

$$q^{E}(B', Y^{G}) = \beta^{E}(1 - \delta(B', Y^{G})).$$
(11)

The probability of default $\delta(B', Y^G)$ defines the risk premium that the government has to pay to the investors to compensate them for the default risk. The higher the default probability, the lower is the price at which the government is able to sell bonds to the investors. The government affects this price only through the default risk.

Now we move to the problem of the domestic bondholders, expressed in (4a) - (4b). By using the no-arbitrage condition, we can define the price of domestic government bonds in terms of the price of the domestic private bond l as:

$$q^{D}(B', B'^{D}, Y^{G}) = (1 - \delta(B', Y^{G})) q^{l}(B', B'^{D}, Y^{G}).$$

As noted before, the price of domestic private bonds is not constant (as it was the case for the external bonds), but depends on the total amount of bonds issued by the government, the amount of domestic bonds and the endowment of the government (that is perfectly positively correlated to the domestic bondholders' endowment). By using the aggregate version of the budget constraint of the young domestic investors (4b) ¹¹ and the market clearing conditions for the bond markets, we obtain an expression for the price of domestic private bonds:

$$q^{l}(B', B'^{D}, Y^{G}) = \frac{Y^{D}}{(1 - \delta(B', Y^{G}))B'^{D} + L}.$$
(12)

We can rewrite the price of domestic bonds as:

$$q^{D}(B', B'^{D}, Y^{G}) = \frac{(1 - \delta(B', Y^{G}))Y^{D}}{(1 - \delta(B', Y^{G}))B'^{D} + L}.$$
(13)

There are two channels through which the government affects the price of domestic private bonds (12) and domestic government bonds (13), due to its market power. On one hand, an increase in B'^D , everything else equal, affects negatively the price of all the assets in the domestic economy, because the government increases the effective supply of bonds in the domestic market. That implies lower prices for domestic government bonds and domestic private bonds. On the other hand, an increase in the default premium due to a marginal increase in risky government bonds (either B'^D or B'^E), everything else equal, affects positively the price of the domestic private bonds. This is a general equilibrium effect of the default risk premium: the effective total supply of domestic assets that bears no sovereign default risk decreases. The effect on the price of domestic bonds is more complex. In fact, the positive general equilibrium effect is counteracted by the negative effect coming from the additional risk premium to pay to bondholders. Although the net effect is negative, the domestic government bond price turns out to be less sensitive to the default premium. That can allow the government to indulge in more risky borrowing.¹² Focusing on the price of external government bonds (11) and the price of domestic government bonds (13), we notice that they are linked through the default risk premium. The default risk premium increases with the total amount of debt issued. This means that an increase

$$Y^{D} = (1 - \delta(B', Y^{G})) q^{l}(B', B'^{D}, Y^{G}) B'^{D} + q^{l}(B', B'^{D}, Y^{G}) L,$$

where $Y^D = \frac{1-\lambda}{2}y$. Since only young domestic bondholders receive an endowment, the part of the output of the economy that is under the control of the government is equal to $\frac{2\lambda}{1+\lambda}$. The part of the output that is under the control of the young domestic bondholders is equal to $\frac{1-\lambda}{1+\lambda}$.

¹²The risk of government bonds affects also the price of the domestic private bonds because of the government's market power. Gourinchas and Jeanne (2012) mention how this general equilibrium effect of the risk premium can explain the small decrease in the yield on US Treasury bill following the downgrading of the US in 2011.

¹¹The budget constraint of the young domestic bondholders in aggregate form becomes:

in B'^D to a level where the default risk is positive, not only affects the price of domestic bonds, but also the price of external bonds. The reverse is true for an increase in B'^E .¹³

1.4 Counterfactual Domestic Private Bond Price

We define the counterfactual domestic private bond price, q_c^l , in order to understand how large is the government's market power and when it is stronger.

$$q_c^l = \frac{\frac{1-\lambda}{1+\lambda}GDP}{L} \tag{14}$$

This is the ratio of the domestic bondholders' endowment, Y^D , over the amount of private bonds, *L*. We call it a "counterfactual" private bond price, because it would be the prevailing price of private savings instruments in the domestic economy, if the government did not issue any domestic debt but still imposed capital controls. In equilibrium, the government issues domestic debt to take advantage of the high prices (low interest rates), but by doing so, it also decreases them (raises the rates) with each additional domestic bond. The higher is this counterfactual price, the more intense is the scarcity of private savings instruments. The difference between q_c^l and q^f gives a measure of the benefits of market segmentation.

The lower is the level of financial development in the domestic economy, i.e. *L/GDP*, or the smaller is the size of the government, λ , the more intense is the scarcity of private savings instruments. Indeed, the benefits from financially repressing domestic bondholders are larger when they represent a larger part of the population, $1 - \lambda$.

1.5 The Link between Debt Laffer Curves and the Composition of debt

In order to understand what drives the government's issuance decisions, we introduce the endogenous "debt Laffer curves". The debt Laffer curve explains how the government's revenues from issuing debt vary with the amount of debt issued. In order to simplify the exposition, we make three simplifying assumptions with respect to the model that we calibrate in the next section. First, the government's endowment shocks are i.i.d and the endowment for the domestic bondholders is fixed. Thus, the debt revenues, and hence the bond prices, depend only on the total amount of debt issued, B'. Second, the probability

¹³Even if we had a model of discriminatory default, this link would still be there as long as the probability of defaulting on one type of debt was positively correlated with the probability of defaulting on the other type of debt.



Figure 2: We plot the debt Laffer curve under three different cases: when only domestic debt is issued (blue solid line), only external debt is issued (red dashed line), or both are issued in equal share (green dash-dotted line).

of default is exogenous and follows a logistic function.¹⁴ Third, the composition of debt is fixed exogenously. Under these three simplifications, we illustrate how the composition of debt affects the Laffer curve and how that differs from the standard sovereign debt models that only feature external debt.

We consider two polar cases, only external debt or only domestic debt issued, and an equal share of domestic and external debt, as shown in Figure 2.¹⁵ The debt Laffer curve represents the increase in current consumption, hence the revenues, due to borrowing on the *y*-axis and the amount of borrowing on the *x*-axis. The value B^{MIN} , defined in (8), is the value of debt after which the probability of default is positive, hence the government has to pay a risk premium to sell its debt. The values *C*, *D*, *E* represent the maxima of the Laffer curve, as defined in (9), for external debt only, for the equal debt composition and for the domestic debt only, respectively. These are the levels of debt that maximize the government's revenues from issuing debt, for each type of debt composition. The interval between B^{MIN} and each maximizing value is the so-called "risky borrowing region", that

$$Prob(def = 1) = \frac{1}{1 + exp(11 - x/2)}$$

Furthermore, we have assumed the external risk-free rate equal to 4 percent, the domestic bondholders' endowment equal to 15 and the amount of private bonds equal to 3. The choice of the parameter values is simply for better representation.

¹⁵To be precise, depending on the model parameters and the endowment shock that hits the economy, the debt Laffer curve for only domestic debt can be higher or lower than the one for the external debt. The current figure is just for exposition purposes.

¹⁴Following Jeanneret and Souissi (2016), we have assumed that the marginal probability of default follows a logistic distribution. In particular, we have used:

is the region where there is borrowing with positive default probabilities, for each case of debt composition. We know that in equilibrium the government will never choose a level of borrowing above *C*, *D*, *E*, because it can attain the same level of consumption for a lower level of borrowing.

We start by describing the Laffer curve for the case of purely external debt. Until B^{MIN} the increase in current consumption is linear in B' because the probability of default is zero. The marginal gain from borrowing is positive and constant: each additional unit of borrowing adds $(1 + r^z)^{-1}$ unit of consumption. After that point the curve becomes concave, due to the risk premium to be paid to external bondholders. In the risky borrowing region the marginal gain from additional borrowing is positive but decreasing, because the default risk increases. After *C* the marginal gain from borrowing becomes negative: the risk premium more than offsets the additional utility of borrowing one more unit.

The Laffer curve for the case of solely domestic debt, instead, is never linear. Until B^{MIN} the curve is concave, even though the probability of default is zero. In fact, domestic market prices increase in B' because of the government's market power in the domestic financial market. This determines the positive but decreasing marginal gain from borrowing. In the risky borrowing region a positive probability of default exists. Here, the risky borrowing region is larger, as it ends at the value of debt equal to *E*. As we mentioned above, the probability of default has a dual role in affecting the price of domestic government bonds: a positive one and a negative one. The net effect on domestic bond prices is negative, but it is substantial only for high levels of default risk. Hence, the marginal gain from borrowing decreases slowly at the beginning of the risky borrowing region and more quickly later on.

The Laffer curve for an equal share of domestic and external debt lies between the other two curves. In general, we can infer that the higher is the ratio of domestic to total debt, the more concave is the Laffer curve and therefore the faster the marginal gain from borrowing decreases, even without default risk. At the same time, the risky borrowing region is larger, the higher is the ratio of domestic to total debt. The discussion of the debt Laffer curve highlights the importance of actually allowing the government to choose the composition of debt. In our model, when the government chooses the debt composition, it is not only choosing the size of its revenues, but also the size of the risky borrowing region, hence the elasticity of its revenues to the default risk. The concavity of the Laffer curve coming from the government's market power suggests that the total debt arising in equilibrium will be low when the domestic to total debt ratio is high. Moreover, the concavity of the risky borrowing region with domestic debt predicts a higher domestic debt share when the default risk is lower.

The risky borrowing region is the most crucial element of sovereign debt models. In

their literature review, Aguiar and Amador (2014) argue that standard sovereign debt models have problems replicating realistic debt levels, because the price schedule of external government bonds is very elastic to the debt issued. In other words, the Laffer curve for solely external debt features a narrow risky borrowing region. Our model helps to expand Aguiar and Amador (2014)'s argument. The key feature of standard sovereign debt models is that the price of external debt is formed in a competitive financial market. Once we add a domestic financial market, where the government has market power and the default risk works as link to the external market, we are able to make the borrowing region larger. In fact, the price schedule becomes less elastic to the amount of debt issued. The optimal debt composition determines the "optimal" risky borrowing region. That explains why our model is able to reproduce realistic debt levels at low default frequencies even when the punishment from default is only short-lived market exclusion.

2 The Quantitative Relevance of Market Exclusion

Having described the trade-offs of the government's debt issuance choice under simplifying assumptions, we move back to the full model where both debt composition and default are endogenous. This section illustrates that our model resolves the disconnect between sovereign debt levels and financial market exclusion. We are able to reproduce the sovereign debt levels and the composition of debt at reasonable default frequencies, even if market exclusion is the only punishment after default.¹⁶ Importantly, we do that by allowing for financial repression in the domestic economy. We calibrate the model to match the domestic debt levels, external debt levels and the domestic and external default frequency of Argentina, Mexico and Russia. We have chosen these three countries, because they have jointly defaulted on the domestic and external debt at some point throughout their recent history and have used the financial repression policies extensively in the period of interest.

2.1 Calibration

We calibrate the model at an annual frequency, due to the frequency of the dataset on government debt. The model is solved with value function iteration on a discrete grid.¹⁷

In Table 1 we report the parameters that are calibrated and in Table 2 the ones that are used to match debt and default statistics. The coefficient of risk aversion of the government

¹⁶By no means we imply that other costs from defaulting that have been considered in the literature to explain the sovereign debt levels are irrelevant. Our goal is to show the importance of financial market exclusion.

¹⁷The two-dimensional grid consists of 300 points for the debt and 61 points for the endowment. In Section 2.5, we show that our results are not sensitive to the number of points for the endowment grid, unlike the standard sovereign model for external debt.

Table 1: Parameters

Common		Argentina	Mexico	Russia
risk aversion: σ		2	2	2
risk-free rate: r^z		0.04	0.04	0.04
Country specific				
persistence (end.): ρ	log(gdp per capita)	0.72	0.82	0.77
st.dev (end.): σ_{ϵ}	log(gdp per capita)	0.053	0.034	0.057
private bonds : L	liquid liabilities % GDP	19	26	25

Note: Annual frequency. Data coverage: Argentina and Mexico (1970-2010), Russia (1992-2010). Source: The data on GDP per capita are from the International Macroeconomics Dataset belonging to md4stata datasets. The data on liquid liabilities as percentage of GDP are obtained from the financial development database 2016.

is set to 2. The world risk-free rate is set to 4 percent. The stochastic endowment process is estimated from data on GDP per capita for the period 1970-2010. In particular, we assume that the GDP per capita has a log-quadratic trend. After subtracting the log-quadratic trend, we obtain the cyclical component and we fit a log-normal AR(1) process with zero mean. We obtain that Argentina has the least persistent process among the three countries and Mexico has the most persistent one. In addition, Mexico's estimated stochastic process is the least volatile among the three. Argentina and Russia have volatile stochastic processes. The endowment shock process is discretized using the Tauchen and Hussey (1991) method. The parameter *L* that regulates the degree of financial development in the economy is set to capture the liquid liabilities as percentage of GDP, which is a measure of financial depth based on Čihák et al. (2012). The last three parameters, the government's discount factor, β , the size of the government, λ , and the probability of market re-entrance, θ , are chosen to target the domestic debt to GDP, external debt to GDP and the default frequency based on joint defaults on domestic and external debt.

The values of β we obtain are low and in line with the standard sovereign debt models (for example, Aguiar and Gopinath (2006)'s discount factor translated into annual frequency is equal to 0.4). The size of the government, λ , implies that domestic bondholders' savings represent 68 percent, 74 percent and 77 percent of the GDP for Argentina, Mexico and Russia, respectively. Independently of the country chosen, the calibration exercise shows that our model with only market exclusion after default can generate the relevant debt levels and a default frequency close to the empirical levels. Remarkably, it only requires the market exclusion to be between 1 and 1.5 years. This number is at the lower bound of the empirical estimates. Compared to a standard sovereign debt model, such as Aguiar and Gopinath (2006) and Arellano (2008), market exclusion is more costly in our

Table 2: Target moments

	Argentina		Mexico		Russia	
	Data	Model	Data	Model	Data	Model
External Debt %GDP	29	29	24	25	26	26
Domestic Debt %GDP	19	18	19	19	20	20
Default frequency (%)	2	1.1	1.5	1.1	2	1.4
Calibrated parameters						
discount factor: β	0.46		0.48		0.48	
government size: λ	0.19		0.15		0.13	
prob. market re-entrance: θ	1		0.82		0.85	

Note: Annual frequency. Data coverage: Argentina and Mexico (1970-2010), Russia (1992-2010). The parameters β , λ , θ are set to match the external debt, domestic debt and default frequency jointly.

Source: The data on government debt are an extended version of the dataset from Panizza (2008). The calculation of the default frequency is based on the data on joint default events on domestic and external debt reported in Reinhart and Rogoff (2009) for the period 1800-2010.

model, because it implies that the government loses access to segmented markets, one of which offers relatively cheap funding.

2.2 Policy Functions and Simulation

We analyze the optimal behavior of the government. First, we study the default incentives. Then, we discuss the pricing functions. Finally, we focus on the issuance decision in the domestic and external market.¹⁸

In this model the government has two ways of smoothing consumption: defaulting or issuing new debt (after repaying the past debt). Issuing new debt is beneficial for the current consumption, especially when the current endowment is low. But first the government has to repay the previous debt. The higher the past debt, the higher the incentives to default. In fact, defaulting is more appealing when roll over of debt is not possible. The weaker the market power of the government, the more likely it is that the benefits of inter-temporal risk sharing using government debt are outweighed by the benefits of defaulting. These are the trade-offs that the government faces in every period.

Figure 3a reports the value of the government when it has the option to default for high, medium and low endowment levels. The value function is higher for higher levels of endowment, reflecting the lower default incentives. Moreover, it is higher for smaller amounts of total debt to repay. Figure 3b shows the government's optimal defualt choice.

¹⁸For space reasons, we present the policy functions for Argentina. Qualitatively, the policy functions for Mexico and Russia are similar.



Figure 3: On the left side we plot the value that maximizes the utility for the government by choosing among the utility from repaying, V^{nd} , and the utility from defaulting, V^d for low (black line), medium (dashed line) and high (dotted line) endowment. On the right side we plot the optimal default choice for the government for low (black line), medium (dashed line) and high (dotted line) endowment.



Figure 4: External debt price

Figure 5: We plot how the price function for external bonds varies with the amount of external bonds issued, conditional on a fixed amount of domestic bonds issued, for low (black line), medium (dashed line) and high (dotted line) endowment.



(a) Domestic debt price conditional on no external debt issued

(b) Domestic debt price conditional on a fixed amount of domestic debt issued

Figure 6: On the left side we plot how the price function for domestic bonds varies with the amount of domestic bonds issued, conditional on no external debt issued, for low (black line), medium (dashed line) and high (dotted line) endowment. On the right side we plot how the price function for domestic bonds varies with the amount of external bonds issued, conditional on a fixed amount of domestic debt issued, for low (black line), medium (dashed line) endowment.



Figure 7: On the left side we plot the optimal choice of external debt for low (black line), medium (dashed line) and high (dotted line) endowment. On the right side we plot the optimal choice for domestic debt for low (black line), medium (dashed line) and high (dotted line) endowment.



Figure 8: On the left side we plot the optimal choice of total debt for low (black line), medium (dashed line) and high (dotted line) endowment. On the right side we plot the optimal choice of domestic debt share for low (black line), medium (dashed line) and high (dotted line) endowment.

The policy function features a monotonic relationship with respect to the state variables. Default incentives are stronger for lower endowment and higher debt to repay. This happens because the market power of the government is stronger with higher endowment, since the domestic demand for bonds is higher. Default is more coslty when the market power is stronger. The persistence of the shocks makes the default incentives even stronger in bad times, because the endowment is more likely to remain low in the following period, if it is low in the current period. Tomz and Wright (2013) report evidence that defaults happen mostly during recessions, hence this behavior is in line with the data.

In Figure 4 we plot the price function of external government bonds across external bond issuance, keeping the domestic debt issuance fixed. The pricing function is a straight line for low levels of debt because the default risk is zero and the external risk-free rate is taken as given by the small open economy. The function becomes convex when the risk premium becomes positive. The price falls to zero when the probability of default is one. This decrease in price is slower and starts later, the higher is the endowment level. This happens because default incentives are weaker for high endowment levels, as shown above. Figure 6a and 6b report the pricing function for domestic government bonds. Since the price of domestic debt depends both on the total amount of debt issued and on the domestic debt share, we fix one dimension per time. In Figure 6a we plot the price of domestic government bonds versus the domestic debt issuance, in case that only domestic debt is issued. As mentioned in the previous section, the pricing function is convex and the slope becomes steeper when the risk premium becomes positive. The price falls to zero when the probability of default is one. Moreover, the price is higher and decreases more slowly, the higher is the endowment level. Hence, the price of domestic debt is monotonic with respect to the endowment level. This happens for two reasons. First, the inelastic demand for assets from domestic bondholders is procyclical, implying that the price of all assets in the domestic economy is higher for higher endowment levels. Second, the probability of default is lower for higher endowment levels, thus making the general equilibrium effect of the risk premium stronger. Figure 6b reports the price of domestic bonds versus the external debt issuance, keeping the domestic issuance fixed. The pricing function is affected by the amount of external debt only through the default risk. Hence, we isolate the effect of the default risk on the pricing function q^{D} . The price is higher and decreases more slowly for higher endowment levels. Again we notice that the price function of domestic debt is monotonic in the endowment level, unlike the price function of external debt.

Figure 7a shows the policy function for the issuance of external debt. The amount of external debt issued does not feature a monotonic relationship with the endowment,

conditional on the amount to be repaid. Actually, it is monotonic with respect to the endowment shock only when the amount to be repaid is sufficiently high. This is in line with what we found for the pricing function of external debt. Figure 7b reports the optimal borrowing decision for the government in the domestic market. The amount of debt issued does not vary significantly with the amount of debt to repay, except at the point where the government defaults and the debt goes to zero. In contrast to the external debt issuance, the domestic debt issuance is monotonic in the endowment level, as is its pricing function. The reason for this result is the tension between using debt for consumption smoothing in bad times while having more market power in good times.

Table 3: Business of	cycle statistics
----------------------	------------------

	Arg	entina	Mexico		Russia	
	Data	Model	Data	Model	Data	Model
Mean						
External Spread (%)	15.2	1.1	3.4	1.1	7.1	1.6
Domestic Spread (%)	NA	0.6	NA	0.6	NA	0.9
Domestic debt share (%)	40	38	44	43	44	44
Standard deviation						
External Spread (%)	17.6	0.6	2.5	0.6	1.1	0.8
Domestic Spread (%)	NA	0.3	NA	0.4	NA	0.5
Correlation with GDPp.c.						
External Spread (%)	-0.3	-0.3	-0.4	-0.2	-0.4	-0.4
Domestic Spread (%)	NA	0	NA	0	NA	0
External debt % GDP	-0.7	-0.8	-0.2	0.7	-0.7	-0.1
Domestic debt % GDP	-0.6	-0.2	0.2	0.5	0.6	0.3
Trade balance % GDP	-0.4	-0.4	0.04	-0.3	NA	-0.5

Note: Annual frequency. Data coverage: Argentina and Mexico (1970-2010), Russia (1992-2010). Source: The data on government debt are an extended version of the dataset from Panizza (2008). The model is simulated for 5,000,000 periods. We extract 50 periods before a default event and calculate the business cycle statistics as the average of all 50-period business cycle statistics averages.

The data about external spreads are taken from Aguiar et al. (2016). The sample coverage is 1993-2014, due to the availability of EMBI data. The data about domestic spreads are not available for the period of interest.

The data about the trade balance are taken from Schmitt-Grohe and Uribe (2017). The sample coverage is 1980-2012, but the data for Russia are not available for a long enough time span at annual frequency.

Having presented the policy functions for domestic debt and external debt, we proceed with the policy function for the optimal domestic debt share. Figure 8b shows that, conditional on the debt to be repaid, the domestic debt share increases with the endowment. This result comes from two driving forces. First, the market power of the government is higher, the higher is the endowment level. This is related to the assumption of procyclical inelastic demand for assets for domestic bondholders. In fact, we have discussed above that the price of domestic debt is monotonic in the endowment level. There is also a second importance force. When the endowment is low, the government has a greater need to issue debt for insurance purposes and issuing large amounts of debt affects the prices more in the domestic market than in the external market. This comes from the market power of the government in the domestic market and it is especially true when the risk of default is low or null.¹⁹

Hence, we obtain a higher domestic debt share when the endowment is higher, that is when default incentives are lower. Interestingly, we obtain this result without invoking any preference of the government for its domestic bondholders. It is simply due to the interaction of the government's market power and the consumption smoothing motive. In contrast, papers that focus on the internal costs of default, such as Mallucci (2015) and Perez (2015), rely on the government's preference for its domestic bondholders. Our mechanism is more relevant to explain the levels and the composition of debt for non-advanced economies, because it is in line with empirical evidence on the use of financial repression in these countries.

Having analyzed the optimal government's behavior, in Table 3 we look at some business cycle statistics. We show which features of the business cycle our model can capture, given that we have only allowed for market exclusion as default punishment and we have matched the debt levels. In terms of the average spread, the model accounts for a small fraction of the value in the data, although it is close to match the default frequency. This shortcoming of the model is common even among sovereign debt models that feature output costs during default and risk-neutral lenders. As explained in Schmitt-Grohe and Uribe (2017), these models cannot match the (external) default frequency and the external spread at the same time.²⁰ However, it is worth noting that our model is able to match the levels of debt, the default frequency and explain one fourth and one third of the external debt spread for Russia and Mexico, respectively.

Moreover, our model is consistent with the evidence, presented in Du and Schreger (2016), regarding the spreads of local currency and foreign currency debt for 10 emerging countries between 2005 and 2014. After extracting the currency risk from local currency

¹⁹In Section 4 we explain that the procyclical domestic debt share, conditional on the amount of debt to repay, is a feature of our model, independently from the assumption of procyclical domestic demand for assets.

²⁰A solution proposed to match the average external spread was to allow the international bondholders to be risk-averse. However, Schmitt-Grohe and Uribe (2017) show that risk aversion under reasonable assumptions and parametrization does not solve the problem.

bond prices, the authors are able to compare the default risk premia of domestic and external bonds and find that the average domestic spread is lower than the average external spread.Du and Schreger (2016) mention sovereign debt market segmentation as potential cause for the gap between the local currency debt and the external currency debt. We obtain this regularity, even if the default in our model is not discriminatory. That is a consequence of the government's market power in the domestic economy. Even if the default probability is the same, since in the domestic economy the government affects the price of the domestic private bonds, *L*, we obtain different spreads across the two markets. The model's performance is relatively weak compared to the data when we look at the standard deviation of the spreads. Indeed, the difficulty of matching the volatility of the spread is typical of sovereign debt models that do not assume asymmetric output costs during the default period, as mentioned in Aguiar et al. (2016).²¹ Taking into account that, it is worth noting that our results are not that far from the data, as long as Mexico and Russia are concerned. On the other hand, the model captures particularly well the correlations of these variables with respect to GDP.

2.3 Segmented versus Integrated Debt Markets - Model Performance

To highlight the importance of incorporating the government's market power in a sovereign debt model, in Table 4 we compare our model's performance with an economy where the government does not exploit its market power. This model is essentially similar to the one studied by Aguiar and Gopinath (2006) and Arellano (2008). Thus, we consider the standard external sovereign debt model and assume that the only punishment after default is the short-lived market exclusion that was obtained in the calibration in section 2.1. Under this assumption, an economy with integrated financial markets can sustain approximately zero debt in equilibrium. Instead, once we allow for the government to take advantage of its market power in the domestic economy, we can match the sovereign debt levels as we have seen in the previous section. Thus, the study of external sovereign debt is linked to the study of domestic sovereign debt and a relevant punishment after default, such as market exclusion, when it entails losing the benefits from domestic marker power.

²¹It was believed that asymmetric default costs during default were sufficient for the model to match the volatility of the external debt spreads in the data. However, Aguiar et al. (2016) show that an additional assumption is needed: a volatile endowment process. In fact, when they calibrate the model to Mexico, that features a less volatile output than Argentina, they are not able to match the volatility of the spread.

Table 4: Segmented and Non segmented debt markets.

		Model			
	Data	Segmented	Non segmented		
Mean					
External Debt % GDP	29	29	approx. 0		
External Spread (%)	15.2	1.1	approx. 0		

Note: Annual frequency. Data coverage: Argentina (1970-2010).

Source: The data on external govenrment debt are an extended version of the dataset from Panizza (2008).

2.4 Default Events - Output

We study the behavior of the economy around default events. In particular, we are interested in the behavior of output around the Argentinian default, because it has been studied extensively in the sovereign debt literature. The model predicts that a default happens at the end of the output contraction, consistently with the evidence (Schmitt-Grohe and Uribe, 2017). In addition, the size of the fall in output at the point of default is within the relevant empirical estimates. On average, in our calibration, during the year of default output falls by 13%, as it is shown in Figure 9. In the data GDP per capita fell by 20% (our own calculation). We conclude that defaults in our model happen at the end of a sharp fall in output. That is important, because the model abstracts from mechanisms that induce a fall in output during default in an exogenous or endogenous manner. This result verifies that market exclusion becomes a relevant punishment, when the government loses access to segmented debt markets.



Figure 9: Output around default events, 10 years before and 10 years after. The model with Argentinean calibration is simulated 5,000,000 periods, then we extract 10 periods before and after a default event and calculate the mean and the standard deviation across default episodes.

Table 5: Sensitivity analysis

	Baseline		в	,	λ	θ	Ygrid
		$\beta_l = 0.41$	$\beta_{h} = 0.51$	$\lambda_l = 0.14$	$\lambda_h = 0.24$	$\theta_l = 0.8$	$n_y = 200$
Default frequency (%)	1.1	1.1	0.9	0.8	1.6	1.0	<u> </u>
Mean							
External Debt (%)	29	34	25	59	13	41	29
Domestic Debt (%)	18	18	18	20	16	18	18
External Spread (%)	1.1	1.2	1.0	0.9	1.8	1.0	1.2
Domestic Spread (%)	0.6	0.6	0.5	0.4	1.0	0.5	0.6
Domestic debt share (%)	38	34	42	25	56	30	38
Standard deviation							
External Spread (%)	0.6	0.6	0.7	0.4	1.2	0.5	0.6
Domestic Spread (%)	0.3	0.3	0.3	0.2	0.7	0.3	0.3
Correlation with GDPp.c.							
External Spread (%)	-0.3	-0.1	-0.3	-0.2	-0.3	-0.3	-0.3
Domestic Spread (%)	0	0	0	0	0	0	0
External debt % GDP	-0.8	-0.9	-0.7	-0.9	0	-0.9	-0.8
Domestic debt % GDP	-0.2	0.1	-0.1	-0.3	0.1	0.1	0.1
Trade balance % GDP	-0.4	-0.4	-0.3	-0.4	-0.3	-0.4	-0.4

Note: The model is simulated for 5,000,000 periods. We extract 50 periods before a default event and calculate the business cycle statistics as the average of all 50-period business cycle statistics averages. In the baseline case we use $\beta = 0.46$, $\lambda = 0.19$, $\theta = 1$ and $n_y = 61$.

2.5 Sensitivity Analysis

In this section we show how sensitive are the results of our model to the value of key parameters. In Table 5 we show how the main moments we obtained for Argentina change when we vary the value of the government's discount factor, β , the size of the government, λ , and the probability of market re-entrance after default, θ . In Appendix B, we show also the sensitivity to the persistence of the output process, ρ , and the variance of the output shock, σ_{ϵ} .

Regarding the government's discount factor, a more impatient government would care less about the cost of default, hence default incentives are higher, everything else equal. Lower β should imply higher default probability. What happens to the level of debt is less clear. On one hand, higher default incentives mean that debt may be more expensive. On the other hand, a more impatient government would like to consume more today, hence has a stronger desire for debt. What we obtain from our exercise is indeed an increase in the default probability for higher β and a higher external debt level. The government is able to satisfy its desire for more debt.²² A lower value for λ implies more domestic and

²²In a similar sensitivity exercise, Schmitt-Grohe and Uribe (2017) obtain a lower debt level, in a model that contains also output costs during default. This comparison shows that the presence of two debt instruments,

external debt and lower default probability. In fact, this corresponds to the case of a larger share of captive domestic bondholders, hence the benefits from market power are larger for the government. When the cost from defaulting is higher, the level of debt that can be sustained is higher. A lower probability of re-entrance in the market after default, θ , means that default is more costly. This implies lower default incentives. In fact, we obtain lower default probability and higher external debt levels. This result shows how important is the market exclusion in our model. In standard sovereign debt models, the role of market exclusion is marginal, as described in Schmitt-Grohe and Uribe (2017). Instead, we obtain a remarkable difference (12% of total debt over GDP) just by increasing the average exclusion from the market from one year to one year and one quarter.

One additional test we perform is to increase the number of points for the endowment grid that is used to solve the model numerically. As first pointed out by Hatchondo, Martinez and Sapriza (2010) and more recently by Schmitt-Grohe and Uribe (2017), the number of grid points used for solving the model with a discrete state space technique matters for the results obtained with standard sovereign debt models. In particular, the average spread and the volatility of the spread are mostly affected. This is due to the sensitivity of the bond price function to the endowment shock, hence the size of the risky borrowing region. Our model is not subject to this problem. Increasing the number of endowment grid points from 61 to 200 does not have an impact on the business cycle statistics. Hence, this is another way to demonstrate that the risky borrowing region in our model is larger than in the standard sovereign debt models. In other words, we have made the bond price function less sensitive to the endowment shock.

3 Financial Repression and the Composition of Sovereign Debt

In this second part of the paper we turn the discussion to the role of financial repression in explaining the domestic and external sovereign debt of non-advanced economies. First, we conduct a calibration exercise in which the domestic and external debts unravel, by only targeting the total debt levels of non-advanced economies. Afterwards, we show that we can obtain negative real interest rates on sovereign debt that are close to the evidence for non-advanced economies during 1966-2010 (Escolano, Shabunina and Woo, 2016).

traded in segmented markets, allows the government to issue more debt when it is more impatient, like in models with full commitment.

Table 6:	Parameters	for	cross-group	calibration

Co	mmon			
risk aversion: σ	standard	2		
risk-free rate: r^z	standard (RBC)	0.04		
discount factor: β	low default frequency	0.75		
persistence (end.): ρ	cross-country average	0.68		
st.dev (end.): σ_{ϵ}	cross-country average	0.042		
prob. market re-entrance: θ	Schmitt-Grohe and Uribe (2017)	0.08		
Group-specific: Wo	rld Bank Income Group	LI	LMI	UMI
private bonds : L	liquid liabilities % GDP	25	39	45
Calibrate	ed parameters	LI	LMI	UMI
government's size: λ	Target: Total debt % GDP	0.265	0.145	0.115

Note: Annual frequency. Data coverage: The data on annual quadratically detrended GDP per capita, liquid liabilities as percentage of GDP and total debt as percentage of GDP, cover the period 1970-2010 conditional on the country and the period belonging to the extended version of the dataset presented in Panizza (2008) and the availability of the data.

Source: The data on GDP per capita are from the International Macroeocnomics Dataset belonging to md4stata datasets. The data on liquid liabilities as percentage of GDP are obtained from the financial development database 2016 and the data on total debt as percentage of GDP are from an extended version of the dataset presented in Panizza (2008).

3.1 Across Income Groups Calibration

The calibration exercise is conducted in the following way. We split the countries according to the World Bank income group they belong to. All the parameters, except the amount of domestic private bonds and the government's size, are common across the groups (see table 6). The amount of domestic private bonds, L, is set to match the average level of financial development for each income group. The measure of financial development is liquid liabilities over GDP. Liquid liabilities as percentage of GDP is considered a measure of financial development related to financial institutions depth, based on the classification in Cihák et al. (2012). The parameter λ , that represent the government's size, is set to match the average total debt over GDP for each income group. Before proceeding with the calibration results, we explain how we set the other parameters that are common across income groups. The coefficient of risk aversion, σ , and the risk-free rate, r^{z} , are the same as in the previous calibration exercise. The choice of β is motivated by the fact that the average default frequency for the countries in our dataset is low. Based on our calculations from the cross-country data described in Reinhart and Rogoff (2009), we find that the frequency of joint defaults on domestic and external debt is around 0.3 percent. Thus, in order to target this low default frequency, β should be equal to 0.75. Then, we turn to to the parameters



(a) Total debt against domestic and external debt for Non-Advanced economies



Figure 10: Domestic and External Sovereign Debt: Model vs Data

Source: Own calculations based on Ugo Panizza's dataset described in Panizza (2008) and the Financial Development dataset 2016.

Description: With hollow circle is the average domestic debt as percentage of GDP for each income group. With solid circle the value of domestic debt as percentage of GDP obtained in our model for each income group. With hollow diamond is the average external debt as percentage of GDP for each income group. With solid diamond the value of external debt as percentage of GDP obtained in our model for each income group.

that characterize the persistence and the standard deviation of the endowment process. We follow the same approach as in the individual country calibration. After obtaining the parameters for each country, we calculate the cross-country average. That implies a value of 0.68 for the persistence and 0.042 for the standard deviation. The probability of re-entrance to the market is set to 0.08. Based on Schmitt-Grohe and Uribe (2017), table 13.6, defaulted countries obtain partial re-access to the financial market after 9.8 years and full re-access to the financial market after 15.9. So we choose the market exclusion to be the average of the two, which is 13 years.

To match the data on total debt over GDP, the model requires more market power for the government for higher total debt levels. We can also derive the counterfactual domestic bond price for each income group, as defined in Formula 14. This measure decreases with the level of economic development: it goes from 2.32 for Low Income countries to 1.76 for Upper-Middle Income countries. In Figure 10 we plot the results from the two non-targeted moments of this calibration exercise, the average domestic and external debt. Remarkably, they closely resemble the data.

The Upper-middle Income group has a more balanced composition of debt compared to the other two groups. According to our model, the reason is that, given the smaller market power for the government for this income group, the benefits from financial repression are low. This implies that market segmentation is less beneficial, which yields a more balanced composition of government debt. On the other side, the Low Income group can sustain high levels of sovereign debt, because the benefits of financial repression are high, which implies that access to segmented debt markets is more valuable compared to the other two income groups. The strong government's market power implies that, everything else equal, one unit of domestic debt is more valuable than one unit of external debt. However, the additional net marginal benefit from domestic debt issuance. Thus, the government issues a small amount of domestic debt that is valuable and a larger amount of external debt.

3.2 Interest Rate Growth Differential

Reinhart and Sbrancia (2015) report that advanced economies reduced their government debt levels after World War II mostly by paying negative real interest rates. Recently, Escolano, Shabunina and Woo (2016) have calculated the interest rate growth differential for advanced and non-advanced economies for the period 1966-2010. The non-advanced economies paid on average -6.1 percent as real interest rate on their sovereign debt, adjusted for growth and currency valuations. Reinhart and Sbrancia (2015) and Escolano, Shabunina and Woo (2016) connect the low interest rates in these countries with the presence of financial repression. Our model of segmented debt markets serves as the best laboratory for explaining the evidence. There is no growth in our model. So, the interest rate growth differential from our model coincides with the real interest rate on sovereign debt. Following Escolano, Shabunina and Woo (2016), the real interest rate on sovereign debt is a weighted average of the real interest rate paid on domestic debt and the real interest rate paid on external debt and the weight is given by their share:

$$R = dds * R^D + (1 - dds) * R^E.$$
⁽¹⁵⁾

 R^{D} refers to the real interest paid on domestic debt, R^{E} refers to the real interest paid on external debt and *dds* is the domestic debt share. Then, taking a weighted average for the three income groups (the weight is given by the number of countries in each income group divided by the total number of countries), we obtain a real interest rate of -3.4 percent (Table 7).

Therefore, the model does not only replicate the levels and composition of debt, but is also consistent with the real interest rate paid on average by non-advanced economies, which is negative. That being said, it implies that the study of sovereign debt should not Table 7: The real interest rate on sovereign debt

Data Model Real Interest rate -6.1% -3.4%

The data on the real interest rate growth differential are based on Escolano, Shabunina and Woo (2016) for the period 1966-2010. The model counterpart is based on our calculations taking a weighted average across income groups. The weight is the fraction of countries belonging to one group out of the total countries that are part of the dataset on sovereign debt that is used in the paper.

refrain from the study of domestic debt for an additional reason. It cannot be consistent with the evidence on real interest rate on government debt, that is negative due to financial repression.

4 Discussion of Assumptions

In this section we discuss the role of two main assumptions. The first one is the procyclical inelastic demand for assets from the domestic bondholders. The second one is the non-discriminatory default.

The inelastic demand for assets from the domestic bondholders is procyclical, because their endowment is (perfectly) positively correlated to the government's endowment. It is reasonable to assume that the resources of all agents in the economy are positively correlated. The fact that they are perfectly correlated is a simplifying, but not crucial, assumption that allows us to carry on only one state variable for the endowment process. In the model, across different calibrations, we obtain a procyclical domestic debt share, conditional on the amount of debt to repay.²³ The procyclical demand for assets makes the market power of the government stronger in good times. Solving the model with a different assumption, either fixed endowment or countercyclical endowment for the domestic bondholders, that would imply acyclical or countercyclical market power, we can still obtain a procyclical domestic debt share, conditional on the amount of debt issued. Instead, the default incentives change, becoming stronger in good times. The reason is that the government's market power is not anymore stronger in good times, hence default is not anymore too costly in good times. We conclude that the main driver behind the procyclical domestic debt share is the higher demand for external debt in bad times, that is obtained depending on the calibration of the government's patience and the output process. Overall, given that

²³However, the unconditional correlation between the output and the domestic debt share can be positive or negative.

the evidence shows that default happens mostly in bad times, we consider our assumption of procyclical demand for assets, and therefore stronger market power in good times, the appropriate one.

The model assumes non-discriminatory default. This assumption allows us to carry over only one state variable for total debt. Moreover, the evidence on whether and how default discrimination happens is inconclusive. As mentioned before, the key mechanism in our model is related to the fact that the government can segment its debt market and possesses market power in the domestic segment. The two debt markets are related through the default risk. Hence, what we really need for our mechanism to work is the default risk of one debt to be positively correlated to the default risk of the other debt. If we had to relax the assumption of non-discrimination, it would be possible to obtain this kind of positive correlation. Our assumption is, therefore, not too stringent.

5 Conclusion

We have provided a model to explain the domestic and external sovereign debt in the presence of limited legal enforcement of sovereign debt contracts. Our starting point is that non-advanced economies are not able to produce enough private savings instruments for "local savers". Hence, there is a role for domestic public debt, especially when the government restricts outward capital flows to exercise market power. The debt market is divided into domestic and external segments and the default risk links the debt instruments issued in the two markets. We have shown that a short-lived market exclusion after default, a previously thought mild default punishment, is enough to match the observable debt levels of Argentina, Mexico and Russia at realistic default frequencies. Moreover, we obtain a composition of debt that resembles the data. This renders the idea of allowing the government to segment its debt market quantitatively relevant.

An additional key result is the connection of domestic and external sovereign debt levels of non-advanced economies with financial repression, consistently with the evidence. We have conducted a quantitative exercise in which we have targeted the total sovereign debt level of the non-advanced economies, by varying the potential for financial repression in the economy. The exercise regards the non-advanced economies, split into the three World Bank income groups they belong to. The model predicts that the higher are the benefits of financial repression - hence the higher the government's market power - the higher are the sustainable debt levels. As a consequence of the differences in the market power across the three income groups, we predict that governments of poorer countries can sustain higher sovereign debt levels, mostly external debt, because the benefits from financially repressing part of the population are higher. Thus, financial repression is not only a tool to explain the domestic sovereign debt levels, as is usually argued, but, most importantly, to explain the external sovereign debt levels as well. Finally, our model rationalizes the average negative real interest rates paid by non-advanced economies on sovereign debt. Domestic and external sovereign debt are inherently linked in a model of financial repression.

In the model we have abstracted from other features of sovereign debt to be able to highlight the link between domestic and external debt. We leave the issue of the maturity structure, debt renegotiation, discriminatory default, currency composition for future research. In addition, it would be interesting to see how market exclusion from segmented debt markets compete with additional costs from defaulting, once we have shown that it is costlier than what was thought. These features would help to improve the model's performance in terms of business cycle statistics.

References

- Abiad, Abdul, Enrica Detragiache, and Thierry Tressel. 2008. "A New Database of Financial Reforms." International Monetary Fund IMF Working Papers 08/266.
- Acharya, Viral V., and Raghuram G. Rajan. 2013. "Sovereign debt, government myopia, and the financial sector." *Review of Financial Studies*, 26(6): 1526–1560.
- **Aguiar, Mark, and Gita Gopinath.** 2006. "Defaultable debt, interest rates and the current account." *Journal of International Economics*, 69(1): 64–83.
- Aguiar, Mark, and Manuel Amador. 2014. "Sovereign debt." Handbook of International Economics, 4: 647–687.
- Aguiar, M., S. Chatterjee, H. Cole, and Z. Stangebye. 2016. "Chapter 21 Quantitative Models of Sovereign Debt Crises." In . Vol. 2 of *Handbook of Macroeconomics*, , ed. John B. Taylor and Harald Uhlig, 1697 – 1755. Elsevier.
- Aizenman, Joshua. 2004. "Financial Opening and Development: Evidence and Policy Controversies." American Economic Review, 94(2): 65–70.
- Arellano, Cristina. 2008. "Default Risk and Income Fluctuations in Emerging Economies." *American Economic Review*, 98(3): 690–712.
- **Chari, V.V., Alessandro Dovis, and Patrick J. Kehoe.** 2016. "On the Optimality of Financial Repression." Federal Reserve Bank of Minneapolis Research Department Staff Report n.1000.

- Chatterjee, Satyajit, Dean Corbae, Makoto Nakajima, and Jose-Victor Rios-Rull. 2007. "A Quantitative Theory of Unsecured Consumer Credit with Risk of Default." *Econometrica*, 75(6): 1525–1589.
- Čihák, Martin, Aslı Demirgüç-Kunt, Erik Feyen, and Ross Levine. 2012. "Benchmarking financial systems around the world." The World Bank Policy Research Working Paper Series 6175.
- Clerc, Laurent, Alexis Derviz, Caterina Mendicino, Stephane Moyen, Kalin Nikolov, Livio Stracca, Javier Suarez, and Alexandros P. Vardoulakis. 2015. "Capital Regulation in a Macroeconomic Model with Three Layers of Default." *International Journal of Central Banking*, 11(3): 9–63.
- **Cole, Harold L., and Patrick J. Kehoe.** 1998. "Models of Sovereign Debt: Partial Versus General Reputations." *International Economic Review*, 39(1): pp. 55–70.
- **D'Erasmo, Pablo, and Enrique G Mendoza.** 2016. "Optimal domestic (and external) sovereign default." National Bureau of Economic Research.
- D'Erasmo, P., E.G. Mendoza, and J. Zhang. 2016. "Chapter 32 What is a Sustainable Public Debt?" In . Vol. 2 of *Handbook of Macroeconomics*, , ed. John B. Taylor and Harald Uhlig, 2493 2597. Elsevier.
- **Dooley, Michael P.** 1995. "A Survey of Academic Literature on Controls over International Capital Transactions." National Bureau of Economic Research, Inc NBER Working Papers 5352.
- **Du, Wenxin, and Jesse Schreger.** 2016. "Local Currency Sovereign Risk." *The Journal of Finance*, 71(3): 1027–1070.
- Eaton, Jonathan, and Mark Gersovitz. 1981. "Debt with Potential Repudiation: Theoretical and Empirical Analysis." *Review of Economic Studies*, 48(2): 289–309.
- **Escolano, Julio, Anna Shabunina, and Jaejoon Woo.** 2016. "The Puzzle of Persistently Negative Interest Rate-Growth Differentials: Financial Repression or Income Catch-Up?" *Fiscal Studies*.
- Farhi, Emmanuel, and Jean Tirole. 2012. "Bubbly Liquidity." *Review of Economic Studies*, 79(2): 678–706.
- Fernández, Andrés, Alessandro Rebucci, and Martín Uribe. 2015. "Are capital controls countercyclical?" *Journal of Monetary Economics*, 76: 1–14.

- **Forslund, Kristine, Lycia Lima, and Ugo Panizza.** 2011. "The determinants of the composition of public debt in developing and emerging market countries." *Review of Development Finance*, 1(3-4): 207 222.
- **Gennaioli, Nicola, Alberto Martin, and Stefano Rossi.** 2014. "Sovereign Default, Domestic Banks, and Financial Institutions." *The Journal of Finance*, 69(2): 819–866.
- **Giovannini, Alberto, and Martha de Melo.** 1993. "Government Revenue from Financial Repression." *American Economic Review*, 83(4): 953–63.
- **Gordon, Roger H., and Wei Li.** 2003. "Government as a discriminating monopolist in the financial market: the case of China." *Journal of Public Economics*, 87(2): 283–312.
- **Gourinchas, Pierre-Olivier, and Olivier Jeanne.** 2012. "Global safe assets." Bank for International Settlements BIS Working Papers 399.
- **Guidotti, Mr Pablo Emilio, and Mr Manmohan S Kumar.** 1991. Domestic Public Debt of Externally Indebted Countries. Occasional Paper n.80, International Monetary Fund.
- Hatchondo, Juan Carlos, Leonardo Martinez, and Horacio Sapriza. 2010. "Quantitative properties of sovereign default models: solution methods matter." *Review of Economic Dynamics*, 13(4): 919–933.
- Jeanneret, Alexandre, and Slim Souissi. 2016. "Sovereign defaults by currency denomination." Journal of International Money and Finance, 60: 197–222.
- Kirkegaard, J, C Reinhart, and M. B. Sbrancia. 2011. "Financial Repression Redux." Finance & Development, 48(1).
- Lucas, Roberto. 1985. "Models of business cycles." Yrjo Jahnsson Lectures. Oxford: Basil Blackwell.
- Mallucci, Enrico. 2015. "Domestic Debt and Sovereign Defaults." International Finance Discussion Papers 1153.
- **Menaldo, Victor.** 2016. "The Fiscal Roots of Financial Underdevelopment." *American Journal* of *Political Science*, 60(2): 456–471.
- **Obstfeld, Maurice.** 1993. "International Capital Mobility in the 1990s." National Bureau of Economic Research, Inc NBER Working Papers 4534.
- **Panizza, Ugo.** 2008. "Domestic and external public debt in developing countries." United Nations Conference on Trade and Development Discussion Paper 188.

- **Perez, Diego.** 2015. "Sovereign debt, domestic banks and the provision of public liquidity." *Manuscript, New York University.*
- **Reinhart, Carmen M., and Kenneth S. Rogoff.** 2009. *This Time Is Different: Eight Centuries of Financial Folly.* Vol. 1 of *Economics Books*, Princeton University Press.
- Reinhart, Carmen M., and Kenneth S. Rogoff. 2011. "The Forgotten History of Domestic Debt." *Economic Journal*, 121(552): 319–350.
- Reinhart, C. M., and M. B. Sbrancia. 2015. "The liquidation of government debt." *Economic Policy*, 30(82): 291–333.
- Schmitt-Grohe, Stephanie, and Martin Uribe. 2017. Open Economy Macroeconomics. Princeton University Press.
- **Tauchen, George, and Robert Hussey.** 1991. "Quadrature-Based Methods for Obtaining Approximate Solutions to Nonlinear Asset Pricing Models." *Econometrica*, 59(2): 371–96.
- **Tomz, Michael, and Mark L.J. Wright.** 2013. "Empirical Research on Sovereign Debt and Default." *Annual Review of Economics*, 5(1): 247–272.

A Data

Table 8 shows the countries included in the analysis, according to the World Bank Income group they belong to. With respect to Panizza (2008)'s dataset we have excluded Comoros, Montenegro, Serbia and Somalia for lack of data on domestic and total debt. We have excluded Iraq, Lithuania, Namibia, Syrian Arab Republic and United Arab Emirates for lack of data on the composition of sovereign debt. We have also excluded Sao Tome and Principe for limited length of data series and Saudi Arabia for lack of data.

Figure 11 reports the distribution of the correlation between the domestic-to-external debt ratio and the detrended GDP for the countries in the dataset.



Figure 11: Distribution of correlation between detrended GDP and Domestic to External debt ratio for non-advanced economies.

Source: own calculations based on Ugo Panizza's dataset described in Panizza (2008).

B Additional Sensitivity Analysis

In this section we provide additional sensitivity analysis. We vary the variance of the output shock, σ_{ϵ} , and the persistence of the output process, ρ .

An increase in the variance of the output process implies that larger shocks hit the government's economy, increasing the incentives to default. On the other hand, the incentives for precautionaty savings increase, and should lead to lower debt issued in equilibrium. In line with these mechanisms and with Schmitt-Grohe and Uribe (2017), we obtain higher default probability, higher spreads and lower external debt when we increase σ_{ϵ} . The do-

Low Income	Lower-Middle Income	Upper-Middle income
Bangladesh	Albania	Algeria
Benin	Angola	Argentina
Burkina Faso	Armenia	Belarus
Burundi	Azerbaijan	Bosnia Herzegovina
Cambodia	Belize	Botswana
Central Africa	Bhutan	Brazil
Chad	Bolivia	Bulgaria
Eritrea	Cameroon	Chile
Ethiopia	Cape Verde	Colombia
Gambia	China	Costa Rica
Ghana	Congo Republic	Dominica
Guinea	Cote d'Ivoire	Dominican Republic
Guinea Bissau	Djibouti	Fiji
Haiti	Ecuador	Gabon
Kenya	Egypt	Grenada
Kyrgyz Republic	El Salvador	Jamaica
Laos	Georgia	Kazakhstan
Liberia	Guatemala	Latvia
Madagascar	Guyana	Lebanon
Malawi	Honduras	Macedonia
Mali	India	Malaysia
Mauritania	Indonesia	Mauritius
Mozambique	Iran	Mexico
Myanmar	Jordan	Panama
Nepal	Lesotho	Peru
Niger	Maldives	Poland
Rwanda	Moldova	Russia
Senegal	Mongolia	Seychelles
Sierra Leone	Morocco	South Africa
Tajikistan	Nicaragua	St Vincent Grens
Tanzania	Nigeria	StKitts and Nevis
Togo	Pakistan	StLucia
Uganda	Papua New Guinea	Turkey
Uzbekistan	Paraguay	Uruguay
Vietnam	Phillipines	Venezuela
Yemen	Samoa	
Zambia	Solomon Islands	
Zimbabwe	Sri Lanka	
	Sudan	
	Swaziland	
	Thailand	
	Tonga	
	Tunisia	

Table 8: List of countries

Turkmenistan

Ukraine Vanuatu

Table 9: Sensitivity analysis - additional

	Baseline		0	C	τ_{ϵ}
		$\rho_l = 0.67$	$ ho_h = 0.77$	$\sigma_{\epsilon}^{l} = 0.048$	$\sigma^h_\epsilon = 0.058$
Default frequency (%)	1.1	0.9	1.4	0.9	1.3
Mean					
External Debt (%)	29	33	27	32	26
Domestic Debt (%)	18	18	18	18	18
External Spread (%)	1.1	1.0	1.5	0.9	1.4
Domestic Spread (%)	0.6	0.5	0.8	0.5	0.7
Domestic debt share (%)	38	35	40	36	40
Standard deviation					
External Spread (%)	0.6	0.6	0.6	0.6	0.6
Domestic Spread (%)	0.3	0.3	0.3	0.3	0.3
Correlation with GDPp.c.					
External Spread (%)	-0.3	-0.2	-0.5	-0.2	-0.3
Domestic Spread (%)	0	0	0	0	0
External debt % GDP	-0.8	-0.9	-0.3	-0.8	-0.8
Domestic debt % GDP	-0.2	0	0	0	0.1
Trade balance % GDP	-0.4	-0.3	-0.4	-0.3	-0.4

The model is simulated for 5,000,000 periods. We extract 50 periods before a default event and calculate the business cycle statistics as the average of all 50-period business cycle statistics averages. In the benchmark case we use $\rho = 0.72$, $\sigma_{\epsilon} = 0.053$.

mestic debt is unaffected by the change in the parameter, because there are other forces at work in the domestic market, due to the market power of the government.

Increasing the persistence of the output process should lead to an increase in the default probability, because bad shocks to the economy are supposed to be followed more likely by more bad shocks. This increases the incentives to default and hence the default premia, making debt more costly. In fact, we obtain a higher default probability and higher spreads on the debt when we increase ρ , while external debt drops.

Earlier Working Papers:

For a complete list of Working Papers published by Sveriges Riksbank, see www.riksbank.se

Estimation of an Adaptive Stock Market Model with Heterogeneous Agents by Henrik Amilon	2005:177
Some Further Evidence on Interest-Rate Smoothing: The Role of Measurement Errors in the Output Gap by Mikael Apel and Per Jansson	2005:178
Bayesian Estimation of an Open Economy DSGE Model with Incomplete Pass-Through by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2005:179
Are Constant Interest Rate Forecasts Modest Interventions? Evidence from an Estimated Open Economy DSGE Model of the Euro Area by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2005:180
Inference in Vector Autoregressive Models with an Informative Prior on the Steady State by Mattias Villani	2005:181
Bank Mergers, Competition and Liquidity by Elena Carletti, Philipp Hartmann and Giancarlo Spagnolo	2005:182
Testing Near-Rationality using Detailed Survey Data by Michael F. Bryan and Stefan Palmqvist	2005:183
Exploring Interactions between Real Activity and the Financial Stance by Tor Jacobson, Jesper Lindé and Kasper Roszbach	2005:184
Two-Sided Network Effects, Bank Interchange Fees, and the Allocation of Fixed Costs by Mats A. Bergman	2005:185
Trade Deficits in the Baltic States: How Long Will the Party Last? by Rudolfs Bems and Kristian Jönsson	2005:186
Real Exchange Rate and Consumption Fluctuations follwing Trade Liberalization by Kristian Jönsson	2005:187
Modern Forecasting Models in Action: Improving Macroeconomic Analyses at Central Banks by Malin Adolfson, Michael K. Andersson, Jesper Lindé, Mattias Villani and Anders Vredin	2005:188
Bayesian Inference of General Linear Restrictions on the Cointegration Space by Mattias Villani	2005:189
Forecasting Performance of an Open Economy Dynamic Stochastic General Equilibrium Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2005:190
Forecast Combination and Model Averaging using Predictive Measures by Jana Eklund and Sune Karlsson	2005:191
Swedish Intervention and the Krona Float, 1993-2002 by Owen F. Humpage and Javiera Ragnartz	2006:192
A Simultaneous Model of the Swedish Krona, the US Dollar and the Euro by Hans Lindblad and Peter Sellin	2006:193
Testing Theories of Job Creation: Does Supply Create Its Own Demand? by Mikael Carlsson, Stefan Eriksson and Nils Gottfries	2006:194
Down or Out: Assessing The Welfare Costs of Household Investment Mistakes by Laurent E. Calvet, John Y. Campbell and Paolo Sodini	2006:195
Efficient Bayesian Inference for Multiple Change-Point and Mixture Innovation Models by Paolo Giordani and Robert Kohn	2006:196
Derivation and Estimation of a New Keynesian Phillips Curve in a Small Open Economy by Karolina Holmberg	2006:197
Technology Shocks and the Labour-Input Response: Evidence from Firm-Level Data by Mikael Carlsson and Jon Smedsaas	2006:198
Monetary Policy and Staggered Wage Bargaining when Prices are Sticky by Mikael Carlsson and Andreas Westermark	2006:199
The Swedish External Position and the Krona by Philip R. Lane	2006:200

Price Setting Transactions and the Role of Denominating Currency in FX Markets by Richard Friberg and Fredrik Wilander	2007:201
The geography of asset holdings: Evidence from Sweden by Nicolas Coeurdacier and Philippe Martin	2007:202
Evaluating An Estimated New Keynesian Small Open Economy Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2007:203
The Use of Cash and the Size of the Shadow Economy in Sweden by Gabriela Guibourg and Björn Segendorf	2007:204
Bank supervision Russian style: Evidence of conflicts between micro- and macro-prudential concerns by Sophie Claeys and Koen Schoors	2007:205
Optimal Monetary Policy under Downward Nominal Wage Rigidity by Mikael Carlsson and Andreas Westermark	2007:206
Financial Structure, Managerial Compensation and Monitoring by Vittoria Cerasi and Sonja Daltung	2007:207
Financial Frictions, Investment and Tobin's q by Guido Lorenzoni and Karl Walentin	2007:208
Sticky Information vs Sticky Prices: A Horse Race in a DSGE Framework by Mathias Trabandt	2007:209
Acquisition versus greenfield: The impact of the mode of foreign bank entry on information and bank lending rates by Sophie Claeys and Christa Hainz	2007:210
Nonparametric Regression Density Estimation Using Smoothly Varying Normal Mixtures by Mattias Villani, Robert Kohn and Paolo Giordani	2007:211
The Costs of Paying – Private and Social Costs of Cash and Card by Mats Bergman, Gabriella Guibourg and Björn Segendorf	2007:212
Using a New Open Economy Macroeconomics model to make real nominal exchange rate forecasts by Peter Sellin	2007:213
Introducing Financial Frictions and Unemployment into a Small Open Economy Model by Lawrence J. Christiano, Mathias Trabandt and Karl Walentin	2007:214
Earnings Inequality and the Equity Premium by Karl Walentin	2007:215
Bayesian forecast combination for VAR models by Michael K. Andersson and Sune Karlsson	2007:216
Do Central Banks React to House Prices? by Daria Finocchiaro and Virginia Queijo von Heideken	2007:217
The Riksbank's Forecasting Performance by Michael K. Andersson, Gustav Karlsson and Josef Svensson	2007:218
Macroeconomic Impact on Expected Default Freqency by Per Åsberg and Hovick Shahnazarian	2008:219
Monetary Policy Regimes and the Volatility of Long-Term Interest Rates by Virginia Queijo von Heideken	2008:220
Governing the Governors: A Clinical Study of Central Banks by Lars Frisell, Kasper Roszbach and Giancarlo Spagnolo	2008:221
The Monetary Policy Decision-Making Process and the Term Structure of Interest Rates by Hans Dillén	2008:222
How Important are Financial Frictions in the US and the Euro Area by Virginia Queijo von Heideken	2008:223
Block Kalman filtering for large-scale DSGE models by Ingvar Strid and Karl Walentin	2008:224
Optimal Monetary Policy in an Operational Medium-Sized DSGE Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Lars E. O. Svensson	2008:225
Firm Default and Aggregate Fluctuations by Tor Jacobson, Rikard Kindell, Jesper Lindé and Kasper Roszbach	2008:226
Re-Evaluating Swedish Membership in EMU: Evidence from an Estimated Model by Ulf Söderström	2008:227

The Effect of Cash Flow on Investment: An Empirical Test of the Balance Sheet Channel by Ola Melander	2009:228
Expectation Driven Business Cycles with Limited Enforcement by Karl Walentin	2009:229
Effects of Organizational Change on Firm Productivity by Christina Håkanson	2009:230
Evaluating Microfoundations for Aggregate Price Rigidities: Evidence from Matched Firm-Level Data on Product Prices and Unit Labor Cost by Mikael Carlsson and Oskar Nordström Skans	2009:231
Monetary Policy Trade-Offs in an Estimated Open-Economy DSGE Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Lars E. O. Svensson	2009:232
Flexible Modeling of Conditional Distributions Using Smooth Mixtures of Asymmetric Student T Densities by Feng Li, Mattias Villani and Robert Kohn	2009:233
Forecasting Macroeconomic Time Series with Locally Adaptive Signal Extraction by Paolo Giordani and Mattias Villani	2009:234
Evaluating Monetary Policy by Lars E. O. Svensson	2009:235
Risk Premiums and Macroeconomic Dynamics in a Heterogeneous Agent Model by Ferre De Graeve, Maarten Dossche, Marina Emiris, Henri Sneessens and Raf Wouters	2010:236
Picking the Brains of MPC Members by Mikael Apel, Carl Andreas Claussen and Petra Lennartsdotter	2010:237
Involuntary Unemployment and the Business Cycle by Lawrence J. Christiano, Mathias Trabandt and Karl Walentin	2010:238
Housing collateral and the monetary transmission mechanism by Karl Walentin and Peter Sellin	2010:239
The Discursive Dilemma in Monetary Policy by Carl Andreas Claussen and Øistein Røisland	2010:240
Monetary Regime Change and Business Cycles by Vasco Cúrdia and Daria Finocchiaro	2010:241
Bayesian Inference in Structural Second-Price common Value Auctions by Bertil Wegmann and Mattias Villani	2010:242
Equilibrium asset prices and the wealth distribution with inattentive consumers by Daria Finocchiaro	2010:243
Identifying VARs through Heterogeneity: An Application to Bank Runs by Ferre De Graeve and Alexei Karas	2010:244
Modeling Conditional Densities Using Finite Smooth Mixtures by Feng Li, Mattias Villani and Robert Kohn	2010:245
The Output Gap, the Labor Wedge, and the Dynamic Behavior of Hours by Luca Sala, Ulf Söderström and Antonella Trigari	2010:246
Density-Conditional Forecasts in Dynamic Multivariate Models by Michael K. Andersson, Stefan Palmqvist and Daniel F. Waggoner	2010:247
Anticipated Alternative Policy-Rate Paths in Policy Simulations by Stefan Laséen and Lars E. O. Svensson	2010:248
MOSES: Model of Swedish Economic Studies by Gunnar Bårdsen, Ard den Reijer, Patrik Jonasson and Ragnar Nymoen	2011:249
The Effects of Endogenuos Firm Exit on Business Cycle Dynamics and Optimal Fiscal Policy by Lauri Vilmi	2011:250
Parameter Identification in a Estimated New Keynesian Open Economy Model by Malin Adolfson and Jesper Lindé	2011:251
Up for count? Central bank words and financial stress by Marianna Blix Grimaldi	2011:252
Wage Adjustment and Productivity Shocks by Mikael Carlsson, Julián Messina and Oskar Nordström Skans	2011:253

Stylized (Arte) Facts on Sectoral Inflation by Ferre De Graeve and Karl Walentin	2011:254
Hedging Labor Income Risk by Sebastien Betermier, Thomas Jansson, Christine A. Parlour and Johan Walden	2011:255
Taking the Twists into Account: Predicting Firm Bankruptcy Risk with Splines of Financial Ratios by Paolo Giordani, Tor Jacobson, Erik von Schedvin and Mattias Villani	2011:256
Collateralization, Bank Loan Rates and Monitoring: Evidence from a Natural Experiment by Geraldo Cerqueiro, Steven Ongena and Kasper Roszbach	2012:257
On the Non-Exclusivity of Loan Contracts: An Empirical Investigation by Hans Degryse, Vasso Ioannidou and Erik von Schedvin	2012:258
Labor-Market Frictions and Optimal Inflation by Mikael Carlsson and Andreas Westermark	2012:259
Output Gaps and Robust Monetary Policy Rules by Roberto M. Billi	2012:260
The Information Content of Central Bank Minutes by Mikael Apel and Marianna Blix Grimaldi	2012:261
The Cost of Consumer Payments in Sweden	2012:262
by Björn Segendorf and Thomas Jansson	
Trade Credit and the Propagation of Corporate Failure: An Empirical Analysis	2012:263
by Tor Jacobson and Erik von Schedvin	
Structural and Cyclical Forces in the Labor Market During the Great Recession: Cross-Country Evidence	2012:264
by Luca Sala, Ulf Söderström and AntonellaTrigari	
Pension Wealth and Household Savings in Europe: Evidence from SHARELIFE	2013:265
by Rob Alessie, Viola Angelini and Peter van Santen	
Long-Term Relationship Bargaining	2013:266
by Andreas Westermark	
Using Financial Markets To Estimate the Macro Effects of Monetary Policy: An Impact-Identified FAVAR*	2013:267
by Stefan Pitschner	
DYNAMIC MIXTURE-OF-EXPERTS MODELS FOR LONGITUDINAL AND DISCRETE-TIME SURVIVAL DATA	2013:268
by Matias Quiroz and Mattias Villani	
Conditional euro area sovereign default risk	2013:269
by André Lucas, Bernd Schwaab and Xin Zhang	
Nominal GDP Targeting and the Zero Lower Bound: Should We Abandon Inflation Targeting?*	2013:270
by Roberto M. Billi	
Un-truncating VARs*	2013:271
by Ferre De Graeve and Andreas Westermark	
Housing Choices and Labor Income Risk	2013:272
by Thomas Jansson	
Identifying Fiscal Inflation [*]	2013:273
by Ferre De Graeve and Virginia Queijo von Heideken	0040.074
On the Redistributive Effects of Inflation: an International Perspective"	2013:274
by Paola Boel	2042.075
Business Cycle Implications of Mongage Spreads	2013:275
by Kall Walefull	2012:276
	2013.270
A detrimental feedback loop: deleveraging and adverse selection	2013-277
hy Christoph Bertsch	2013.211
Distortionary Fiscal Policy and Monetary Policy Goals	2013.278
by Klaus Adam and Roberto M. Billi	2010.210
Predicting the Spread of Financial Innovations: An Epidemiological Approach	2013.279
by Isaiah Hull	_010.210
Firm-Level Evidence of Shifts in the Supply of Credit	2013:280
by Karolina Holmberg	

Lines of Credit and Investment: Firm-Level Evidence of Real Effects of the Financial Crisis by Karolina Holmberg	2013:281
A wake-up call: information contagion and strategic uncertainty by Toni Ahnert and Christoph Bertsch	2013:282
Debt Dynamics and Monetary Policy: A Note	2013:283
by Stefan Laséen and Ingvar Strid	
Optimal taxation with home production	2014:284
by Conny Olovsson	
Incompatible European Partners? Cultural Predispositions and Household Financial Behavior	2014:285
by Michael Haliassos. Thomas Jansson and Yigitcan Karabulut	
How Subprime Borrowers and Mortgage Brokers Shared the Piecial Behavior	2014:286
by Antie Berndt, Burton Hollifield and Patrik Sandås	
The Macro-Financial Implications of House Price-Indexed Mortgage Contracts	2014:287
by Isaiah Hull	
Does Trading Anonymously Enhance Liquidity?	2014:288
by Patrick J. Dennis and Patrik Sandås	
Systematic bailout guarantees and tacit coordination	2014:289
by Christoph Bertsch. Claudio Calcagno and Mark Le Quement	
Selection Effects in Producer-Price Setting	2014:290
by Mikael Carlsson	2011.200
Dynamic Demand Adjustment and Exchange Rate Volatility	2014:291
by Vesna Corbo	
Forward Guidance and Long Term Interest Rates: Inspecting the Mechanism	2014.292
by Ferre De Graeve. Pelin Ilbas & Raf Wouters	2011202
Firm-I evel Shocks and Labor Adjustments	2014.293
by Mikael Carlsson, Julián Messina and Oskar Nordström Skans	2011.200
A wake-up call theory of contagion	2015:294
by Toni Ahnert and Christoph Bertsch	2010.201
Risks in macroeconomic fundamentals and excess bond returns predictability	2015:295
by Rafael B. De Rezende	2010.200
The Importance of Reallocation for Productivity Growth: Evidence from European and US Banking	2015:296
by Jaan W.B. Bos and Peter C. van Santen	2010.200
SPEEDING LIP MCMC BY EFFICIENT DATA SUBSAMPLING	2015:297
by Matias Quiroz Mattias Villani and Robert Kohn	2010.201
Amortization Requirements and Household Indeptedness: An Application to Swedish-Style Mortgages	2015:208
by Isaiah Hull	2013.290
Fuel for Economic Growth?	2015:299
by Johan Gars and Conny Olovsson	2010.200
Searching for Information	2015:300
by Jungsuk Han and Francesco Sangiorgi	2010.000
What Broke First? Characterizing Sources of Structural Change Prior to the Great Recession	2015:301
by Isaiah Hull	2010.001
Price Level Targeting and Risk Management	2015:302
by Roberto Billi	2010.002
Central bank policy paths and market forward rates: A simple model	2015:303
by Ferre De Graeve and Jens Iversen	2010.000
Jump-Starting the Furn Area Recovery: Would a Rise in Core Fiscal Spending Halp the Perinhery?	2015:304
by Olivier Blanchard Christonher I Erreg and Jesper Lindé	2010.004
Bringing Financial Stability into Monatary Policy*	2015.205
by Fric M Leener and James M Nason	2013.305
SCALABLE MCMC FOR LARGE DATA PROBLEMS USING DATA SUBSAMPLING AND	2015:306
by MATIAS QUIROZ. MATTIAS VILLANI AND ROBERT KOHN	

SPEEDING UP MCMC BY DELAYED ACCEPTANCE AND DATA SUBSAMPLING by MATIAS QUIROZ	2015:307
Modeling financial sector joint tail risk in the euro area by André Lucas, Bernd Schwaab and Xin Zhang	2015:308
Score Driven Exponentially Weighted Moving Averages and Value-at-Risk Forecasting	2015:309
by André Lucas and Xin Zhang	
On the Theoretical Efficacy of Quantitative Easing at the Zero Lower Bound	2015:310
by Paola Boel and Christopher J. Waller	
Optimal Inflation with Corporate Taxation and Financial Constraints	2015:311
by Daria Finocchiaro. Giovanni Lombardo. Caterina Mendicino and Philippe Weil	
Fire Sale Bank Recapitalizations	2015:312
by Christoph Bertsch and Mike Mariathasan	
Since you're so rich, you must be really smart: Talent and the Finance Wage Premium	2015:313
by Michael Böhm. Daniel Metzger and Per Strömberg	
Debt. equity and the equity price puzzle	2015:314
by Daria Finocchiaro and Caterina Mendicino	2010.011
Trade Credit: Contract-Level Evidence Contradicts Current Theories	2016:315
by Tore Ellingsen, Tor, Jacobson and Erik von Schedvin	2010.010
Double Lightlity in a Branch Banking System: Historical Evidence from Canada	2016:316
by Appa Grodecka and Aptonic Kotidis	2010.310
Subprime Perrowers, Securitization and the Transmission of Pupinges Cuelos	2016-217
Subplime Borrowers, Securitization and the Transmission of Business Cycles	2010.317
Deel Time Forecesting for Menetery Delieu Anglusia, The Case of Custines Dilyhords	2010:210
kear- Time Forecasting for Monetary Policy Analysis. The Case of Svenges Riksbank	2010.316
by Jens Iversen, Stelan Laseen, Hennik Lundvall and On Soderstrom	2010:210
Fed Liftoff and Supprime Loan Interest Rates: Evidence from the Peer-to-Peer Lending	2016:319
by Christoph Bertsch, Isalah Huli and Xin Zhang	0040-000
Curbing Shocks to Corporate Liquidity: The Role of Trade Credit	2016:320
by Niklas Amberg, Tor Jacobson, Erik von Schedvin and Robert Townsend	
Firms' Strategic Choice of Loan Delinquencies	2016:321
by Paola Morales-Acevedo	
Fiscal Consolidation Under Imperfect Credibility	2016:322
by Matthieu Lemoine and Jesper Lindé	
Challenges for Central Banks' Macro Models	2016:323
by Jesper Lindé, Frank Smets and Rafael Wouters	
The interest rate effects of government bond purchases away from the lower bound	2016:324
by Rafael B. De Rezende	
COVENANT-LIGHT CONTRACTS AND CREDITOR COORDINATION	2016:325
by Bo Becker and Victoria Ivashina	
Endogenous Separations, Wage Rigidities and Employment Volatility	2016:326
by Mikael Carlsson and Andreas Westermark	
Renovatio Monetae: Gesell Taxes in Practice	2016:327
by Roger Svensson and Andreas Westermark	
Adjusting for Information Content when Comparing Forecast Performance	2016:328
by Michael K. Andersson, Ted Aranki and André Reslow	
Economic Scarcity and Consumers' Credit Choice	2016:329
by Marieke Bos, Chloé Le Coq and Peter van Santen	
Uncertain pension income and household saving	2016:330
by Peter van Santen	
Money, Credit and Banking and the Cost of Financial Activity	2016:331
by Paola Boel and Gabriele Camera	
Oil prices in a real-business-cycle model with precautionary demand for oil	2016:332
by Conny Olovsson	
Financial Literacy Externalities	2016:333
by Michael Haliasso, Thomas Jansson and Yigitcan Karabulut	

The timing of uncertainty shocks in a small open economy	2016:334
by Hanna Armelius, Isaiah Hull and Hanna Stenbacka Köhler	
Quantitative easing and the price-liquidity trade-off	2017:335
by Marien Ferdinandusse, Maximilian Freier and Annukka Ristiniemi	
What Broker Charges Reveal about Mortgage Credit Risk	2017:336
by Antje Berndt, Burton Hollifield and Patrik Sandåsi	
Asymmetric Macro-Financial Spillovers	2017:337
by Kristina Bluwstein	
Latency Arbitrage When Markets Become Faster	2017:338
by Burton Hollifield, Patrik Sandås and Andrew Todd	
How big is the toolbox of a central banker? Managing expectations with policy-rate forecasts: Evidence from Sweden	2017:339
by Magnus Åhl	
International business cycles: quantifying the effects of a world market for oil	2017:340
by Johan Gars and Conny Olovsson I	
Systemic Risk: A New Trade-Off for Monetary Policy?	2017:341
by Stefan Laséen, Andrea Pescatori and Jarkko Turunen	
Household Debt and Monetary Policy: Revealing the Cash-Flow Channel	2017:342
by Martin Flodén, Matilda Kilström, Jósef Sigurdsson and Roine Vestman	
House Prices, Home Equity, and Personal Debt Composition	2017:343
by Jieying Li and Xin Zhang	
Identification and Estimation issues in Exponential Smooth Transition Autoregressive Models	2017:344
by Daniel Buncic	



Sveriges Riksbank Visiting address: Brunkebergs torg 11 Mail address: se-103 37 Stockholm

Website: www.riksbank.se Telephone: +46 8 787 00 00, Fax: +46 8 21 05 31 E-mail: registratorn@riksbank.se